RLI CORP Form 4 May 18, 2011

# FORM 4 UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Symbol

RLI CORP [RLI]

(Month/Day/Year)

3. Date of Earliest Transaction

**OMB APPROVAL** 

OMB Number:

3235-0287

Expires: January 31, 2005

2005

Estimated average burden hours per

5. Relationship of Reporting Person(s) to

(Check all applicable)

Issuer

Director

response... 0.5

10% Owner

Check this box if no longer subject to Section 16. Form 4 or

STATEMENT OF CHANGES IN BENEFICIAL OWNERSHIP OF SECURITIES

Form 5
obligations
may continue.
See Instruction

Filed pursuant to Section 16(a) of the Securities Exchange Act of 1934, Section 17(a) of the Public Utility Holding Company Act of 1935 or Section 30(h) of the Investment Company Act of 1940

2. Issuer Name and Ticker or Trading

1(b).

(Last)

(Print or Type Responses)

1. Name and Address of Reporting Person \*

(First)

(Middle)

DONDANVILLE JOSEPH E

9025 N. LI	9025 N. LINDBERGH DRIVE (Street)		2011	Officer (give title Other (specify below)					
	(Street)		nendment, Date Original onth/Day/Year)	6. Individual or Joint/Group Filing(Check Applicable Line) _X_ Form filed by One Reporting Person					
PEORIA, I	L 61615		Form filed by More than One Reporting Person						
(City)	(State)	(Zip) Tal	Table I - Non-Derivative Securities Acquired, Disposed of, or Beneficially Owned						
1.Title of Security (Instr. 3)	2. Transaction Date (Month/Day/Year)  05/17/2011	2A. Deemed Execution Date, if any (Month/Day/Year)	3. 4. Securities Acquired Transaction(A) or Disposed of (D) Code (Instr. 3, 4 and 5) (Instr. 8)  (A) or Code V Amount (D) Price  G V 200 D	5. Amount of Securities Beneficially Owned Following Reported Transaction(s) (Instr. 3 and 4)	6. Ownership Form: Direct (D) or Indirect (I) (Instr. 4)	7. Nature of Indirect Beneficial Ownership (Instr. 4)			
Stock Common Stock	03/17/2011		G V 200 D 59.54	108,262.015 27,351.7305	I	By Empl. Stock Ownership Plan			
Common Stock				10,336.7652	I	By Trust (1)			
Common Stock				12,276	I	By Wife in Trust			

Reminder: Report on a separate line for each class of securities beneficially owned directly or indirectly.

Persons who respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB control number.

SEC 1474

(9-02)

De

Table II - Derivative Securities Acquired, Disposed of, or Beneficially Owned (e.g., puts, calls, warrants, options, convertible securities)

1. Title of Derivative Security (Instr. 3)	2. Conversion or Exercise Price of Derivative Security	3. Transaction Date (Month/Day/Year)	3A. Deemed Execution Date, if any (Month/Day/Year)	4. Transactic Code (Instr. 8)	5. DriNumber of Derivative Securities Acquired (A) or Disposed of (D) (Instr. 3, 4, and 5)	6. Date Exercisab Expiration Date (Month/Day/Year		7. Title and A Underlying S (Instr. 3 and	Securities
				Code V	(A) (D)	Date Exercisable	Expiration Date	Title	Amount or Number of Shares
Stock Option	\$ 37.54 (2)					05/05/2006	05/05/2015	Common Stock	21,000
Stock Option	\$ 43.15 (2)					05/04/2007(3)	05/04/2016	Common Stock	17,500
Stock Option	\$ 49.09 (2)					05/03/2008(3)	05/03/2017	Common Stock	18,000
Stock Option	\$ 43 (2)					05/01/2009(3)	05/01/2018	Common Stock	5,000
Stock Option	\$ 47.36 (2)					08/01/2009(3)	08/01/2018	Common Stock	5,000
Stock Option	\$ 49.73 (2)					11/03/2009(3)	11/03/2018	Common Stock	5,000
Stock Option	\$ 49.89 (2)					02/02/2010(3)	02/02/2019	Common Stock	5,000
Stock Option	\$ 39.9 (2)					05/07/2010(3)	05/07/2017	Common Stock	4,700
Stock Option	\$ 42.9 (2)					11/02/2010(3)	11/02/2017	Common Stock	4,700
Stock Option	\$ 43.49 (2)					08/03/2010(3)	08/03/2017	Common Stock	4,700
Stock Option	\$ 44.62 (2)					02/01/2011(3)	02/01/2018	Common Stock	4,700
						05/06/2011(3)	05/06/2018		4,000

Stock Option	\$ 49.34 (2)		Common Stock	
Stock Option	\$ 49.2 (2)	08/02/2011 <u>(3)</u> 08/02/2018	Common Stock	4,000
Stock Option	\$ 50.26 (2) (3)	11/01/2011 11/01/2018	Common Stock	4,000
Stock Option	\$ 55.28	02/01/2012 <u>(3)</u> 02/01/2019	Common Stock	4,000
Stock Option	\$ 58.73	05/05/2012 <u>(3)</u> 05/05/2019	Common Stock	4,000

# **Reporting Owners**

Reporting Owner Name / Address Relationships

Director 10% Owner Officer Other

DONDANVILLE JOSEPH E 9025 N. LINDBERGH DRIVE PEORIA, IL 61615

Senior Vice President/CFO

# **Signatures**

/s/ Joseph E. Dondanville 05/18/2011

\*\*Signature of Reporting Date
Person

# **Explanation of Responses:**

- \* If the form is filed by more than one reporting person, see Instruction 4(b)(v).
- \*\* Intentional misstatements or omissions of facts constitute Federal Criminal Violations. See 18 U.S.C. 1001 and 15 U.S.C. 78ff(a).
- (1) Ownership reflects dividend reinvestment.
- (2) Stock Option grant price adjusted to reflect \$7 RLI extraordinary dividend paid 12/29/10.
- (3) Pursuant to option schedule wherein 20% of the aggregate number of shares granted may be exercised commencing one year from grant date and each year thereafter in 20% increments.

Note: File three copies of this Form, one of which must be manually signed. If space is insufficient, *see* Instruction 6 for procedure. Potential persons who are to respond to the collection of information contained in this form are not required to respond unless the form displays a currently valid OMB number. TR>

Safety - Lost-time injuries

5 0.0% 5.0% 0.0%

Safety - Aircraft ground damage

1.8 166.7% 5.0% 8.3%

Employee Engagement/Customer Satisfaction

12 months 200.0% 10.0% 20.0%

Reporting Owners 3

CASM (Cost per available seat mile, excluding fuel)\*

7.85 cents 118.3% 10.0% 11.8%

Alaska Air Group Profitability

>\$350 million 200.0% 70.0% 140.0%

Non-Ticket Passenger Revenue

\$11.55 8.0%

Total Payout %

188.2%

Payout as a % of Base Salary

= 141.1%

Participation Rate\*\*

x 75.0%

- \* Based on Alaska Airlines performance.
- \*\* Participation rates vary based by position. The participation rate used in this example is for one of the NEOs.

52

The Performance-Based Pay plan has paid out as follows since its inception:

In addition, all of the Company s employees, including the executive officers, participate in a separate incentive plan called Operational Performance Rewards, which pays a monthly incentive payment of an equal amount to all employees when certain operational performance targets are met. Awards are based on the achievement of on-time performance and customer satisfaction goals, and the maximum annual payout for each employee is \$1,200.

# Long-Term Equity-Based Pay

Long-term equity-based incentive awards that link executive pay to stockholder value are an important element of the Company s executive compensation program. Long-term equity incentives that vest over three- or four-year periods are awarded annually, resulting in overlapping vesting periods. The awards are designed in such a way as to discourage short-term risk-taking and are primarily intended to align Named Executive Officers long-term interests with those of stockholders. In addition, equity-based awards help attract and retain top-performing executives who fit a team-oriented and performance-driven culture.

Stock Options: The Company makes a portion of its long-term incentive grants to Named Executive Officers in the form of stock options with an exercise price that is equal to the fair market value of the Company's common stock on the grant date. Thus, the Named Executive Officers will realize value from their stock options only to the degree that Air Group stockholders would realize value if they purchased shares and held them for the same period the executive holds his or her stock options. The stock options also function as a retention incentive for executives as they generally vest ratably over the four-year period after the date of grant.

Table of Contents 5

53

Restricted Stock Units: The Company also grants long-term incentive awards to Named Executive Officers in the form of restricted stock units. Subject to the executive s continued employment with the Company, the restricted stock units generally vest on the third anniversary of the date they are granted and, upon vesting, are paid in shares of Alaska Air Group common stock. Thus, the units are designed to link executives interests with those of Air Group s stockholders (as the units value is based on the value of Air Group common stock) and to provide a long-term retention incentive through the vesting period.

**Performance Stock Units:** The Company also grants the Named Executive Officers performance stock units as part of the long-term equity-based incentive program. The performance stock units vest only if the Company achieves performance goals established by the Committee for the performance period covered by the award. Beginning in 2010, performance stock units are tied to total shareholder return (TSR) as compared to an industry peer group. In combination with other stock unit awards described above, the performance stock unit awards are designed to provide an incentive to achieve specific performance goals important to the Company s success.

The performance stock units granted in 2010 are eligible to vest based on the Company's total shareholder return ("TSR") relative to the following peer group over the three-year period commencing January 1, 2010: AirTran Holdings, AMR, Continental Airlines, Delta Air Lines, ExpressJet Holdings, JetBlue Airways, Hawaiian Holdings, Mesa Air Group, Republic Airways Holdings, SkyWest, Southwest Airlines, United Airlines and US Airways Group. (The Committee will adjust the peer group annually as it deems appropriate if one or more of the peer airlines ceases to be a publicly traded company.) The Committee chose TSR as a performance measure for these awards to provide additional incentive for executives to help create shareholder value. Given the nature of the airline business, the Committee believes that measuring TSR on a relative basis against an industry peer group rather than on an absolute basis provides a more relevant reflection of the Company's performance due to macro-economic factors that tend to affect the entire industry and that are largely not under the control of executives. The percentage of the performance stock units that vest may range from 0% to 200% of the target number of units subject to the award, depending on the Company's relative TSR for the performance period.

**Vesting of Prior Performance Grants:** In February 2008, the Company granted awards of performance stock units to Messrs. Ayer, Johnson, Tilden, Minicucci, Pinneo, and Pedersen. The performance units subject to these awards were eligible to vest based on the Company s adjusted pre-tax profit margin over the three-year period commencing on January 1, 2008, with the number of units eligible to vest ranging from 0% to 200% of the target number of units subject to the awards.

At the end of 2010, the Committee determined that 75% of the target number of units subject to each executive s award vested based on the Company s adjusted pre-tax profit margin of 5.3% over the performance period.

*Equity Guidelines*: The Committee considers and generally follows equity grant guidelines that are determined based on the target total direct compensation levels and pay mix described above. Target equity grants, when combined with all other compensation elements described above, are designed to achieve total direct compensation at the 50th percentile of the peer group data for Named Executive Officers. The Committee

54

may adjust equity grants to the Named Executive Officers above or below these target levels based on the Committee s general assessment of:

the individual s contribution to the success of the Company s financial performance;

internal pay equity;

the individual s performance of his job responsibilities; and

the accounting impact to the Company and potential dilution effects of the grant.

Generally, the Committee balances the value of equity incentive compensation awards equally among stock options, time-based restricted stock units and performance stock units.

2010 Equity Awards: For 2010, the guidelines applied to the Named Executive Officers are noted in the table below:

# **Equity Award Guidelines**

Name	Equity Target as % of Base Pay	Stock Options	Equity Mix Restricted Stock Units	Performance Stock Units
William S. Ayer	300%	34%	33%	33%
Glenn S. Johnson	200%	34%	33%	33%
Bradley D. Tilden	250%	34%	33%	33%
Benito Minicucci	200%	34%	33%	33%
Jeffrey D. Pinneo	200%	34%	33%	33%
Brandon S. Pedersen	75%/100%*	34%	33%	33%

<sup>\*</sup> Pro-rated as a result of mid-year election as chief financial officer.

Special Equity Awards: The Committee retains discretion to make other equity awards at such times and on such terms as it considers appropriate to help achieve the goals of the Company's executive compensation program. Mr. Johnson received a grant of additional performance stock units in 2010 in connection with the scope of responsibility he assumed during 2010 to lead the transformation of Company subsidiary Horizon Air Industries. These performance stock units are tied to the achievement of specific financial and operational goals within a certain time period, thereby providing an incentive to strengthen Horizon as a business, which will benefit stockholders of Alaska Air Group.

# Stock Ownership Guidelines

In 2009, the Company adopted stock ownership guidelines for elected officers. Under the guidelines, elected officers are strongly encouraged to hold Company stock having a value of at least one year s base salary. The Committee reviews executive ownership annually.

Perquisites and Personal Benefits

Beginning in 2008, an annual amount equal to 12% of base salary is paid to each Named Executive Officer in lieu of all perquisites except for travel, life insurance, health exams, accidental death and dismemberment insurance and Alaska Airlines Boardroom membership. In addition, the Company will provide lifetime travel to Mr. Ayer as part of his retirement.

55

# Retirement Benefits/Deferred Compensation

The Company provides retirement benefits to the Named Executive Officers under the terms of qualified and non-qualified defined-benefit retirement plans. The Retirement Plan for Salaried Employees (the Salaried Retirement Plan ) and the Company s 401(k) plans are tax-qualified retirement plans that the Named Executive Officers participate in on substantially the same terms as other participating employees. Due to maximum limitations imposed by the Employee Retirement Income Security Act of 1974 and the Internal Revenue Code on the annual amount of a pension which may be paid under a qualified defined benefit plan, the benefits that would otherwise be payable to the Named Executive Officers under the Salaried Retirement Plan are required to be limited. An unfunded defined benefit plan, the 1995 Elected Officers Supplementary Retirement Plan (the Supplementary Retirement Plan ), provides make-up benefits plus supplemental retirement benefits.

The Named Executive Officers are also permitted to elect to defer up to 100% of their annual Performance-Based Pay payments under the Company s Nonqualified Deferred Compensation Plan. The Company believes that providing the Named Executive Officers with deferred compensation opportunities is a cost-effective way to permit executives to receive the tax benefits associated with delaying the income tax event on the compensation deferred.

Please see the 2010 Pension Benefits and 2010 Nonqualified Deferred Compensation tables and information following them for a description of these plans.

# **Agreements Regarding Change in Control and Termination**

The Company has change-in-control agreements with the Named Executive Officers that provide for severance benefits if the executive s employment terminates under certain circumstances in connection with a change in control.

The Company has entered into change-in-control agreements with these executives because it believes that the occurrence, or potential occurrence, of a change-in-control transaction would create uncertainty and disruption during a critical transaction time for the Company. The payment of cash severance benefits under the agreements is triggered if two conditions are met: (1) actual or constructive termination of employment and (2) a change-in-control transaction. The Committee believes that Named Executive Officers should be entitled to receive cash severance benefits only if both conditions are met. Once the change-in-control event commences, the Named Executive Officer s severance and benefits payable under the contract begins to diminish with time, until ultimate expiration of the agreement 36 months later.

# **Policy with Respect to Section 162(m)**

Section 162(m) of the Internal Revenue Code generally prohibits the Company from deducting certain compensation over \$1 million paid to its CEO and certain other executive officers unless such compensation is based on performance objectives meeting certain criteria or is otherwise excluded from the limitation. The Committee strives whenever possible to structure its compensation plans such that they are tax deductible, and it believes that a substantial portion of compensation paid

56

under its current program (including the annual incentives, performance stock units and stock option grants described above) satisfies the requirements under Section 162(m). However, the Committee reserves the right to design programs that recognize a full range of performance criteria important to its success, even where the compensation paid under such programs may not be deductible. For 2010, the Company believes that no portion of its tax deduction for compensation paid to its Named Executive Officers will be disallowed under Section 162(m).

# COMPENSATION COMMITTEE REPORT(1)

The Compensation Committee has certain duties and powers as described in its charter. The Compensation Committee is currently composed of the three non-employee directors named at the end of this report, each of whom is independent as defined by the NYSE listing standards.

The Compensation Committee has reviewed and discussed with management the disclosures contained in the Compensation Discussion and Analysis section of this proxy statement. Based upon this review and discussion, the Compensation Committee recommended to our Board of Directors that the Compensation Discussion and Analysis section be included in the Company s 2010 Annual Report on Form 10-K on file with the SEC and the Company s 2011 Proxy Statement.

**Compensation Committee of the Board of Directors** 

Phyllis J. Campbell, Chair

Jessie J. Knight, Jr., member

Dennis F. Madsen, member

- J. Kenneth Thompson, member
- (1) SEC filings sometimes incorporate information by reference. This means the Company is referring you to information that has previously been filed with the SEC and that this information should be considered as part of the filing you are reading. Unless the Company specifically states otherwise, this report shall not be deemed to be incorporated by reference and shall not constitute soliciting material or otherwise be considered filed under the Securities Act or the Exchange Act.

# COMPENSATION COMMITTEE INTERLOCKS AND INSIDER PARTICIPATION

The Compensation Committee members whose names appear on the Compensation Committee Report above were committee members during all of 2010 except for Mr. Thompson and Mr. Knight, each of whom served a partial year as a result of Mr. Thompson replacing Mr. Knight as a committee member in June 2010. No member of the Compensation Committee during 2010 is or has been an executive officer or employee of the Company or has had any relationships requiring disclosure by the Company under the SEC s rules requiring disclosure of certain relationships and related-party transactions. None of the Company s executive officers served as a director or a member of a compensation committee (or other committee serving an equivalent function) of any other entity, the executive officers of which served as a director or member of the Compensation Committee during the fiscal year ended December 31, 2010.

57

# 2010 SUMMARY COMPENSATION TABLE

The following table presents information regarding compensation of the CEO, the two individuals who served as CFO and the three other most highly compensated executive officers for services rendered during 2010. These individuals are referred to as Named Executive Officers in this Proxy Statement.

Name and Principal				Stock	Option	Non-Equity Incentive Plan	Change in Pension Value and Nonqualified Deferred Compensation	All Other	
Position	<b>X</b> 7	Salary	Bonus	Awards	Awards	Compensation	Earnings	Compensation	Total
(a)	Year (b)	(\$) (c)	(\$) (d)	(\$)(1) (e)	(\$)(1) (f)	(\$)(2) (g)	(\$)(3) (h)	(4) (i)	(\$) ( <b>j</b> )
William S. Ayer	2010	395,385	0	1,120,197	697,052	745,314	305,617	93,785	3,357,350
Chair, President and CEO	2009	373,846	0	1,549,782	1,675,171	621,585	271,949	78,424	4,570,757
	2008	360,000	0	927,100	646,441	186,840	194,473	68,155	2,383,009
Glenn S. Johnson(5)	2010	299,999	0	1,450,732	176,880	421,269	351,001	68,889	2,768,770
President (Horizon Air)	2009	311,537	0	388,651	419,948	388,863	417,941	61,656	1,988,596
Former CFO	2008	277,462	0	393,107	268,126	108,002	637,129	49,556	1,733,382
Bradley D. Tilden	2010	370,961	0	578,724	274,345	594,627	319,527	80,522	2,218,706
President (Alaska Airlines)	2009	353,074	0	544,145	590,016	440,577	250,643	81,315	2,259,770
	2008	278,538	0	489,322	335,464	108,421	57,324	55,181	1,324,250
Benito Minicucci(6)	2010	280,961	0	352,556	167,856	397,776	0	118,663	1,317,812
Executive Vice President/	2009	259,610	0	305,916	333,248	324,215	18,487	64,642	1,306,118
Operations and COO									
(Alaska Airlines)									
Jeffrey D. Pinneo(7)	2010	239,942	0	299,340	140,782	333,380	201,682	67,304	1,282,430
Former President and CEO (Horizon Air)	2009	247,558	0	234,260	252,036	340,366	176,615	71,032	1,321,867
	2008	237,000	0	335,378	226,499	78,032	150,068	63,103	1,090,080
Brandon S. Pedersen(8)	2010	219,389	0	198,010	46,205	269,578	0	89,003	821,185
Vice President/Finance and CFO									

(1)

The amounts reported in Columns (e) and (f) of the Summary Compensation Table above reflect the fair value of these awards on the grant date as determined under the principles used to calculate the value of equity awards for purposes of the Company s financial statements (disregarding any estimate of forfeitures related to service-based vesting conditions). No stock awards or option awards granted to Named Executive Officers were forfeited in any of the three years presented. For a discussion of the assumptions and methodologies used to value the awards reported in Column (e) and Column (f), please see the discussion of stock awards and option awards contained in Note 10 (Stock-Based Compensation Plans) to the Company s Consolidated Financial Statements, included as part of the Company s 2010 Annual Report filed on Form 10-K with the SEC and incorporated herein by reference. For information about the stock awards and option awards granted in 2010 to the Named Executive Officers, please see the discussion under Grants of Plan-Based Awards below.

58

The amounts reported in Column (e) of the table above also include the grant date fair value of performance-based stock unit awards granted in 2008 and 2010 to the Named Executive Officers based on the probable outcome (determined as of the grant date) of the performance-based conditions applicable to the awards. Performance-based stock unit awards were not granted in 2009. The following table presents the aggregate grant date fair value of these awards included in Column (e) for 2008 and 2010 and the aggregate grant date value of these awards assuming that the highest level of performance conditions will be achieved.

	2008 Perfo Aggregate Grant	ormance Awards	2010 Perfo	2010 Performance Awards		
	Date Fair Value (Based on Probable	Aggregate Grant Date Fair Value (Based on Maximum	Aggregate Grant Date Fair Aggregate Value Date Fair (Based on Probable (Based on M			
	Outcome)	Performance)	Outcome)	Performance)		
Name	(\$)	(\$)	(\$)	(\$)		
William S. Ayer	607,529	1,215,058	372,512	745,024		
Glenn S. Johnson	252,908	505,816	1,264,476	1,450,732		
Bradley D. Tilden	316,135	632,270	289,362	578,724		
Benito Minicucci	31,064	62,127	176,278	352,556		
Jeffrey D. Pinneo	214,422	428,844	149,670	299,340		
Brandon S. Pedersen	41,757	83,515	48,892	97,784		

- (2) Non-Equity Incentive Plan Compensation is Performance-Based Pay Plan compensation and Operational Performance Rewards, further described in the Compensation Discussion and Analysis section above.
- (3) The amount reported in Column (h) of the Summary Compensation Table above reflects the year-over-year change in present value of accumulated benefits determined as of December 31 of each year for the Retirement Plan for Salaried Employees and the Officers Supplementary Retirement Plan (defined benefit plan) as well as earnings on each Named Executive Officer s account under the Nonqualified Deferred Compensation Plan. For Mr. Minicucci and Mr. Pedersen, Company contributions to the Defined-Contribution Officers Supplementary Retirement Plan (DC-OSRP) in lieu of the defined-benefit plan are reported in Column (i) and detailed in the table in Footnote (4) below.
- (4) The following table presents detailed information on the types and amounts of compensation reported for the Named Executive Officers in Column (i) of the Summary Compensation Table. For Column (i), each perquisite and other personal benefit is included in the total and identified and, if it exceeds the greater of \$25,000 or 10% of the total amount of perquisites and other benefits for that officer, is quantified in the table below. All reimbursements of taxes with respect to perquisites and other benefits are identified and quantified. Tax reimbursements are provided for travel privileges unique to the airline industry. Also included in the total for Column (i) are: the Company s incremental cost of providing flight benefits; Alaska Airlines Board Room membership; annual physical; and accidental death and dismemberment insurance premiums. By providing positive-space travel without tax consequences to Named Executive Officers, we are able to deliver a highly valued benefit at a low cost to the Company. In addition, we believe that this benefit provides the opportunity for Named Executive Officers to connect with the Company s frontline employees.

Name	Company Contribution to 401(k) Account	Company Contribution to DC-OSRP Account	Term Life Insurance Premiums (and Taxes Paid)	Medical Insurance Paid	Perquisite Allowance	Travel Taxes Paid	Other
William S. Ayer	7,350	N/A	2,977(1,708)	10,728	47,446	21,044	1,987
Glenn S. Johnson(5)	5,365	N/A	1,593(914)	10,687	36,000	11,229	2,557
Bradley D. Tilden	7,350	N/A	1,224(702)	10,728	44,515	13,407	2,050
Benito Minicucci	14,700	45,718	354(203)	10,728	33,554	9,909	2,951
Jeffrey D. Pinneo	16,500	N/A	690(396)	9,843	27,688	10,312	1,331
Brandon S. Pedersen	13,163	25,730	277(159)	10,728	26,327	10,703	1,371

(5) Mr. Johnson was elected President of Horizon Air Industries, Inc. in June 2010. Previously he was Executive Vice President/Finance and CFO of Alaska Air Group, Inc. The Compensation Committee granted a special performance stock unit award to Mr. Johnson upon his election to president of Horizon Air on 6/10/10. In 2008, Mr. Johnson became partially vested under the Supplementary Retirement Plan, and earned sufficient service under the Plan to accrue a benefit payable at his Normal Retirement Age. As a result of these changes, the amount shown in column (h) of the Summary Compensation Table

includes \$574,992 as the Change in Pension Value and is attributable to Mr. Johnson s vesting and service accruals under the Supplementary Retirement Plan during 2008. The Supplementary Retirement Plan and the nonqualified benefits are payable over the long term after Mr. Johnson retires from the Company.

(6) Mr. Minicucci was elected Executive Vice President/Operations and Chief Operating Officer of Alaska Airlines, Inc. in December 2008, and was not a Named Executive Officer prior to 2009. As such, only Mr. Minicucci s 2009 and 2010 compensation data is included.

59

- (7) Mr. Pinneo served as President and CEO of Horizon Air until June 2010, and retired from Horizon Air in January 2011.
- (8) Mr. Pedersen was elected Chief Financial Officer of Alaska Air Group, Inc. and Alaska Airlines, Inc. in June 2010 and was not a Named Executive Officer prior to 2010. As such, only Mr. Pedersen s 2010 compensation data is included.

# 2010 GRANTS OF PLAN-BASED AWARDS

The following table presents information regarding the incentive awards granted to the Named Executive Officers for 2010. Each of the equity-based awards reported in the table below was granted under our 2008 Performance Incentive Plan (2008 Plan).

			ed Possible on-Equity Plan Awards	e Payouts Incentive	Estimated Future Payouts Under Equity Incentive Plan Awards			All Other Stock Awards: Number of Shares	All Other Option Awards: Number of Securities	Exercise or Base Price of	Grant Date Fair Value of Stock and
	,	Threshold	Target	Maximum T	hreshold	Target	Maximum	of	<b>Underlying Options</b>	Option Awards	Option Awards
	Grant	(\$)	(\$)	(\$)	(#)	(#)	(#)	Units	(#)	(\$/Sh)	(\$)(1)
Name	Date (b)	(c)	( <b>d</b> )	(e)	<b>(f)</b>	(g)	(h)	(i)	<b>(j</b> )	(k)	(l)
William S. Ayer Stock Options RSUs PSUs PBP Plan	2/3/10 2/3/10 2/3/10 N/A	98,846	395,385	790,769	2,240	11,200	22,400	22,480	38,620	33.26 0 0	697,052 747,684.80 372,512
Glenn S. Johnson Stock Options RSUs	2/3/10 2/3/10							5,600	9,800	33.26	176,880 186,256
PSUs	2/3/10 6/10/10				1,120	5,600 22,000(2)	11,200	2,000		0	186,256 1,078,220
PBP Plan	N/A	56,250	224,999	449,999							
Bradley D. Tilden  Stock Options RSUs PSUs	2/3/10 2/3/10 2/3/10	70.000	215 215	(20, (24	1,740	8,700	17,400	8,700	15,200	33.26 0 0	274,345 289,362 289,362
PBP Plan	N/A	78,829	315,317	630,634							
Benito Minicucci Stock Options RSUs PSUs PBP Plan	2/3/10 2/3/10 2/3/10 N/A	52,680	210,721	421,441	1,060	5,300	10,600	5,300	9,300	33.26 0 0	167,856 176,278 176,278
Jeffrey D. Pinneo Stock Options RSUs PSUs PBP Plan	2/3/10 2/3/10 2/3/10 N/A	44,989	179,957	359,913	900	4,500	9,000	4,500	7,800	33.26 0 0	140,782 149,670 149,670
Brandon S. Pedersen											

Edgar Filing: RLI CORP - Form 4

Stock Options	2/3/10								2,560	33.26	46,205
RSUs	2/3/10							1,470		0	48,892
	6/10/10							2,045(3)		0	100,225
PSUs	2/3/10				294	1,470	2,940			0	48,892
PBP Plan	N/A	35,651	142,603	285,205							

Key: RSUs Restricted Stock Units; PSUs Performance Stock Units; PBP Plan Performance-Based Pay Plan

60

<sup>(1)</sup> The amounts reported in Column (l) reflect the fair value of these awards on the grant date as determined under the principles used to calculate the value of equity awards for purposes of the Company s financial statements and may or may not be

representative of the value eventually realized by the executive. For a discussion of the assumptions and methodologies used to value the awards reported in Column (1), please see the discussion of stock awards and option awards contained in Note 10 (Stock-Based Compensation Plans) to the Company s Consolidated Financial Statements, included as part of the Company s 2010 Annual Report filed on Form 10-K with the SEC and incorporated herein by reference.

- (2) The Committee granted a special performance stock unit award to Mr. Johnson upon his election to president of Horizon Air on 6/10/10.
- (3) The Committee granted a special restricted stock unit award to Mr. Pedersen upon his election to chief financial officer on 6/10/10.

# **OUTSTANDING EQUITY AWARDS AT FISCAL YEAR-END**

The following table presents information regarding the outstanding equity awards held by each of the Named Executive Officers as of December 31, 2010, including the vesting dates for the portions of these awards that had not vested as of that date.

Name (a) William S. Ayer	Option Grant/ Award Date (b) 11/17/04	Number of Securities Underlying Unexercised Options Exercisable (#) (c) 20,000	Number of Securities Underlying Unexercised Options Unexercisable (#) (d)	Option Exercise Price (\$) (e) 28.85	Option Expiration Date (f) 11/17/14	Number of Shares or Units of Stock That Have Not Vested (#) (g)	Market Value of Shares or Units of Stock That Have Not Vested (\$)(1) (h)	Equity Incentive Plan Awards: Number of Unearned Shares, Units, or Other Rights That Have Not Vested (#) (i)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$)(1) (j)
William S. Ayer	11/17/04 8/30/05 9/13/06 1/31/07 2/8/08 1/29/09 1/29/09 2/3/10 2/3/10	20,000 46,100 37,300 17,850 26,400 12,825 0	0 0 5,950(2) 26,400(6) 38,475(10) 68,338(11) 38,620(12)	28.85 32.96 37.96 42.85 27.49 27.56	11/17/14 8/30/15 9/13/16 1/31/17 2/8/18 1/29/19 2/3/20	11,625(6) 24,125(10) 32,108(11) 22,480(12)	659,021 1,367,646 1,820,203 1,274,391	0 0 0 11,200(13)	0 0 0 634,928
Glenn S. Johnson	1/31/07 6/14/07 2/8/08 1/29/09 2/3/10 2/3/10 6/10/10	0 0 0 0	1,245(2) 2,237(3) 10,950(6) 22,494(10) 9,800(12)	42.85 27.40 27.49 27.56 33.26	1/31/17 6/14/17 2/8/18 1/29/19 2/3/20	5,100(6) 14,102(10) 5,600(12)	289,119 799,442 317,464	0 0 0 5,600(13) 22,000(14)	0 0 0 317,464 1,247,180
Bradley D. Tilden	8/30/05 9/13/06	12,900 11,550	0 0	32.96 37.96	8/30/15 9/13/16				

Edgar Filing: RLI CORP - Form 4

	1/31/07 6/14/07 2/8/08 1/29/09 2/3/10 2/3/10	9,225 5,775 13,700 10,534 0	3,075(2) 1,925(3) 13,700(6) 31,604(10) 15,200(12)	42.85 27.40 27.49 27.56 33.26	1/31/17 6/14/17 2/8/18 1/29/19 2/3/20	6,300(6) 19,744(10) 8,700(12)	357,147 1,119,287 493,203	0 0 0 8,700(13)	0 0 0 493,203
Benito Minicucci	1/31/07 9/19/07 11/6/07	0 0 819	565(2) 282(4) 820(5)	42.85 25.23 23.36	1/31/17 9/19/17 11/6/17				
	2/8/08 6/11/08	0	2,670(6)	27.49	2/8/18	1,630(6) 4,890(7)	92,405 277,214	0 0	0
	6/12/08 12/4/08 1/29/09	0	1,913(8) 17,850(10)	17.88 27.56	6/12/18	20,000(9) 11,100(10)	1,133,800 629,259	0	0
	2/3/10 2/3/10	0	9,300(12)	33.26	2/3/20	5,300(12)	300,457	0 5,300(13)	0 300,457

61

			Option Awards				Stock	x Awards	Equity Incentive Plan Awards:
Name (a)	Option Grant/ Award Date (b)	Number of Securities Underlying Unexercised Options Exercisable (#) (c)	Number of Securities Underlying Unexercised Options Unexercisable (#) (d)	Option Exercise Price (\$) (e)		Number of Shares or Units of Stock That Have Not Vested (#) (g)	Market Value of Shares or Units of Stock That Have Not Vested (\$)(1) (h)	Equity Incentive Plan Awards: Number of Unearned Shares, Units, or Other Rights That Have Not Vested (#) (i)	Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$)(1) (j)
Jeffrey D. Pinneo	8/30/05 9/13/06 1/31/07 2/8/08 1/29/09 2/3/10 2/3/10	10,800 9,780 7,800 0 0	0 0 2,600(2) 9,250(6) 13,500(10) 7,800(12)	32.96 37.96 42.85 27.49 27.56 33.26	8/30/15 9/13/16 1/31/17 2/8/18 1/29/19 2/3/20	4,400(6) 8,500(10) 4,500(12)	249,436 481,865 255,105	0 0 0 4,500(13)	0 0 0 255,105
Brandon S. Pedersen	9/13/06 12/1/06 1/31/07 2/8/08 6/11/08 1/29/09 2/3/10 2/3/10 6/10/10	2,770 2,000 2,940 0 1,750	0 0 980(2) 3,600(6) 5,250(10) 2,560(12)	37.96 39.98 42.85 27.49 27.56 33.26	9/13/16 12/1/16 1/31/17 2/8/18 1/29/19 2/3/10	2,010(6) 5,060(7) 3,290(10) 1,470(12) 2,045(15)	113,947 286,851 186,510 83,334 115,931	0 0 0 0 1,470(13)	0 0 0 0 83,334

- (1) The dollar amounts shown in Column (h) and Column (j) are determined by multiplying the number of shares or units reported in Column (g) and Column (i), respectively, by \$56.69 (the closing price of our common stock on the last trading day of fiscal 2010).
- (2) The unvested options under the 1/31/07 grant will become fully vested on 1/31/11.
- (3) The unvested options under the 6/14/07 grant will become fully vested on 6/14/11.
- (4) The unvested options under the 9/19/07 grant will become fully vested on 9/19/11.
- (5) The unvested options under the 11/6/07 grant will become fully vested on 11/6/11.
- (6) The RSUs awarded on 2/8/08 became fully vested on 2/8/11. The unvested options under the 2/8/08 grant will become vested as follows:

  Mr. Ayer 13,200 on 2/8/11 and 13,200 on 2/8/12; Mr. Johnson 5,475 on 2/8/11 and 5,475 on 2/8/12; Mr. Tilden 6,850 on 2/8/11 and 6,850 on 2/8/12; Mr. Minicucci 1,335 on 2/8/11 and 1,335 on 2/8/12; Mr. Pinneo 4,625 on 2/8/11 and 4,625 on 2/8/12; and Mr. Pedersen 1,800 on 2/8/11 and 1,800 on 2/8/12.
- (7) The RSUs awarded on 6/11/08 become fully vested on 6/11/11.
- (8) The unvested options under the 6/12/08 grant will vest as follows: 956 on 6/12/11 and 957 on 6/12/12.
- (9) The RSUs awarded on 12/4/08 become fully vested on 12/4/11.
- (10) The RSUs awarded on 1/29/09 will become fully vested on 1/29/12. The unvested options under the 1/29/09 grant will become vested as follows: Mr. Ayer 12,825 on 1/29/11; 12,825 on 1/29/12; and 12,825 on 1/29/2013; Mr. Johnson 4,498 on 1/29/11; 7,498 on 1/29/12 and 7,498 on 1/29/13; Mr. Tilden 10,535 on 1/29/11; 10,534 on 1/29/12 and 10,535 on 1/29/13; Mr. Minicucci 5,950 on 1/29/11; 5,950 on 1/29/12 and 5,950 on 1/29/13; Mr. Pinneo 4,500 on 1/29/11; 4,500 on 1/29/12 and 4,500 on 1/29/13; and Mr. Pedersen 1,750 on 1/29/11; 1,750 on 1/29/12 and 1,750 on 1/29/13.
- (11) The awards granted to Mr. Ayer on 1/29/09 will fully vest on the third anniversary of the grant date, or 1/29/12.

(12) The RSUs awarded on 2/3/10 will become fully vested on 2/3/13. The unvested options under the 2/3/10 grant will become vested as follows: Mr. Ayer 9,655 on 2/3/11, 9,655 on 2/3/12, 9,655 on 2/3/13 and 9,655 on 2/3/14; Mr. Johnson 2,450 on 2/3/11, 2,450 on 2/3/12, 2,450 on 2/3/13 and 2,450 on 2/3/14; Mr. Tilden 3,800 on 2/3/11, 3,800 on 2/3/12, 3,800 on 2/3/13 and 3,800 on 2/3/14; Mr. Minicucci 2,325 on 2/3/11, 2,325 on 2/3/12, 2,325 on 2/3/13 and 2,325 on 2/3/14; Mr. Pinneo 1,950 on 2/3/11, 1,950 on 2/3/12, 1,950 on 2/3/13 and 1,950 on 2/3/14; and Mr. Pedersen 640 on 2/3/11, 640 on 2/3/12, 640 on 2/3/13 and 640 on 2/3/14.

62

- (13) The performance stock units reported in Column (i) are eligible to vest based on the Company s performance over a three-year period as described in the Compensation Discussion and Analysis section above and in footnote (1) to the Summary Compensation Table above. The performance stock units granted on 2/3/10 will vest based on the goals set for a three-year performance period ending 12/31/12.
- (14) The performance stock units granted to Mr. Johnson on 6/10/10 will vest based on the goals set for a two-year performance period ending 7/31/12.
- (15) The RSUs awarded on 6/10/10 will become fully vested on 6/10/13.

# 2010 OPTION EXERCISES AND STOCK VESTED

The following table presents information regarding the exercise of stock options by Named Executive Officers during 2010 and on the vesting during 2010 of other stock awards previously granted to the Named Executive Officers.

	Option Number of Shares	on Awards	Stock Awards			
Name	Acquired on Exercise (#)	Value Realized on Exercise (\$)(1)	Number of Shares Acquired on Vesting (#)	Value Realized on Vesting (\$)(1)		
(a)	<b>(b)</b>	(c)	( <b>d</b> )	(e)		
William S. Ayer	289,731	7,198,651	21,775	1,179,016		
Glenn S. Johnson	89,524	1,326,742	7,990	457,131		
Bradley D. Tilden	40,653	861,565	11,825	629,001		
Benito Minicucci	17,483	336,197	1,348	67,652		
Jeffrey D. Pinneo	75,250	1,982,240	8,150	430,687		
Brandon S. Pedersen	13,450	187,751	1,993	96,405		

(1) The dollar amounts shown in Column (c) above for option awards are determined by multiplying (i) the number of shares of our common stock to which the exercise of the option related, by (ii) the difference between the per-share closing price of our common stock on the date of exercise and the exercise price of the options. The dollar amounts shown in Column (e) above for stock awards are determined by multiplying the number of shares or units, as applicable, that vested by the per-share closing price of our common stock on the vesting date.

63

# 2010 PENSION BENEFITS

# **Pension and Other Retirement Plans**

The Company maintains two primary defined benefit pension plans covering Named Executive Officers, other than Mr. Minicucci and Mr. Pedersen. The Alaska Air Group, Inc. Retirement Plan for Salaried Employees (the Salaried Retirement Plan ) is the qualified defined-benefit employee retirement plan, and the Named Executive Officers, other than Mr. Minicucci and Mr. Pedersen, participate in this plan on the same general terms as other eligible employees. The Named Executive Officers, other than Mr. Minicucci and Mr. Pedersen, also participate in the Alaska Air Group, Inc. 1995 Elected Officers Supplementary Retirement Plan (the Supplementary Retirement Plan ).

The following table presents information regarding the present value of accumulated benefits that may become payable to the Named Executive Officers under the qualified and nonqualified defined-benefit pension plans.

Name		Number of Years Credit Service	Present Value of Accumulated Benefit	Payments During Last Fiscal Year
	Plan Name	(#)(1)	(\$)(1)	(\$)
(a)	(b)	(c)	(d)	(e)
William S. Ayer	Salaried Retirement Plan	15.362	636,780	N/A
	Supplementary Retirement Plan	15.398	1,997,557	N/A
Glenn S. Johnson	Salaried Retirement Plan	15.704	498,184	N/A
	Supplementary Retirement Plan	7.431	1,189,574	N/A
Bradley D. Tilden	Salaried Retirement Plan	19.844	594,869	N/A
	Supplementary Retirement Plan	11.919	965,060	N/A
Benito Minicucci(2)	Salaried Retirement Plan	N/A	N/A	N/A
	Supplementary Retirement Plan	N/A	N/A	N/A
Jeffrey D. Pinneo	Salaried Retirement Plan	3.816	25,131	N/A
	Supplementary Retirement Plan	8.920	1,457,786	N/A
Brandon S. Pedersen(2)	Salaried Retirement Plan	N/A	N/A	N/A
	Supplementary Retirement Plan	N/A	N/A	N/A

- (1) The years of credited service and present value of accumulated benefits shown in the table above are presented as of December 31, 2010 assuming that each Named Executive Officer retires at normal retirement age and that benefits are paid out in accordance with the terms of each plan described below. For a description of the material assumptions used to calculate the present value of accumulated benefits shown above, please see Note 6 (Employee Benefits Plans) to the Company s Consolidated Financial Statements, included as part of the Company s 2010 Annual Report filed on Form 10-K with the SEC and incorporated herein by reference.
- (2) In lieu of participation in the defined-benefit plans, Mr. Minicucci and Mr. Pedersen receive an enhanced contribution to the Company s defined-contribution plans. Specifically, in lieu of participation in the qualified defined-benefit pension plan, Mr. Minicucci and Mr. Pedersen each receive a Company match contribution up to 6% of their eligible wages. In lieu of the Supplementary Retirement defined-benefit plan, Mr. Minicucci and Mr. Pedersen participate in the Supplementary Retirement defined-contribution plan, which is further described below.

64

#### Salaried Retirement Plan

The Salaried Retirement Plan is a tax-qualified, defined-benefit retirement plan for salaried Alaska Airlines employees hired prior to April 1, 2003. Each of the Named Executive Officers that participates in the Plan is fully vested in his accrued benefits under the Salaried Retirement Plan. Benefits payable under the Salaried Retirement Plan are generally based on years of credited service with the Company and its affiliates and final average base salary for the five highest complete and consecutive calendar years of an employee s last ten complete calendar years of service. The annual retirement benefit at age 62 (normal retirement age under the Salaried Retirement Plan) is equal to 2% of the employee s final average base salary times years of credited service (limited to 40 years). Annual benefits are computed on a straight life annuity basis beginning at normal retirement age. Benefits under the Salaried Retirement Plan are not subject to offset for Social Security benefits.

The tax law limits the compensation on which annual retirement benefits are based. For 2010, this limit was \$245,000. The tax law also limits the annual benefits that may be paid from a tax-qualified retirement plan. For 2010, this limit on annual benefits was \$195,000.

# Supplementary Retirement Plans

In addition to the benefits described above, the Named Executive Officers, other than Mr. Minicucci and Mr. Pedersen, are eligible to receive retirement benefits under the Supplementary Retirement defined-benefit plan. This plan is a non-qualified, unfunded, defined-benefit plan. Normal retirement benefits are payable once the officer reaches age 60. Benefits are calculated as a monthly amount on a straight life annuity basis. In general, the monthly benefit is determined as a percentage (50% to 75% of a participant s final average monthly base salary) with the percentage determined based on both the officer s length of service with the Company and length of service as an elected officer. This benefit amount is subject to offset by the amount of the officer s Social Security benefits and the amount of benefits paid under the Salaried Retirement defined-benefit plan to the extent such benefits were accrued after the officer became a participant in the Supplementary Retirement defined-benefit plan. (There is no offset for any Salaried Retirement Plan benefits accrued for service before the officer became a participant in the Supplementary Retirement defined-benefit plan.)

Participants in the Supplementary Retirement defined-benefit plan become fully vested in their benefits under the plan upon attaining age 50 and completing 10 years of service as an elected officer. Plan benefits will also become fully vested upon a change of control of the Company or upon termination of the participant semployment due to death or disability.

In lieu of the Supplementary Retirement defined-benefit plan, Mr. Minicucci and Mr. Pedersen participate in the Company s Supplementary Retirement Defined-Contribution plan. Under this plan, the Company contributes 10% of the officer s eligible wages, as defined in plan documents, minus the maximum legal Company contribution that the Company made, or could have made, under the Company s qualified defined-contribution plan.

65

# 2010 NONQUALIFIED DEFERRED COMPENSATION

Under the Deferred Compensation Plan, the Named Executive Officers and other key employees may elect to receive a portion of some or all of their Performance-Based Pay awards on a deferred basis. There were no deferrals in 2010. The crediting interest rate for amounts deferred in prior years is based on the mean between the high and the low rates during the first 11 months of the preceding year of yields of Ba2-rated industrial bonds as determined by the plan administrator (rounded to the nearest one-quarter of one percent). Participants under the plan have the opportunity to elect among the investment funds offered under our 401(k) plan for purposes of determining the return on their plan accounts. Alternatively, participants may allocate some or all of their plan account to an interest-bearing option with a rate equal to the yield on a Moody s index of Ba2-rated industrial bonds as of November of the preceding year, rounded to the nearest one-quarter of one percent. Subject to applicable tax laws, amounts deferred under the plan are generally distributed on termination of the participant s employment, although participants may elect an earlier distribution date and/or may elect payment in a lump sum or installments.

The following table presents information regarding the contributions to and earnings on the Named Executive Officers balances under the Company's nonqualified deferred compensation plans during 2010, and also shows the total deferred amounts for the Named Executive Officers as of December 31, 2010.

Name	Executive Contributions in Last FY	Registrant Contributions in Last FY	Aggregate Earnings in Last FY	Aggregate Withdrawals/ Distributions	Aggregate Balance at Last FYE
ivanie	(\$)	(\$)	(\$)(1)	(\$)	(\$)(1)
(a)	(b)	(c)	(d)	(φ) (e)	(f)
William S. Ayer	0	0	23,437	38,718	306,285
Glenn S. Johnson	0	0	44,566	128,833	556,172
Bradley D. Tilden	0	0	0	0	0
Benito Minicucci	0	45,718	3,033	0	26,196
Jeffrey D. Pinneo	0	1,087	11,906	0	72,392
Brandon S. Pedersen	0	25,730	3,455	0	38,750

(1) Only the portion of earnings on deferred compensation that is considered to be at above-market rates under SEC rules is required to be included as compensation for each Named Executive Officer in Column (h) of the Summary Compensation Table. However, total earnings for each Named Executive Officer listed in Column (d) above were included as earnings in column (h) of the Summary Compensation Table.

66

# POTENTIAL PAYMENTS UPON CHANGE IN CONTROL AND TERMINATION

Under the change-in-control agreements in place with the Named Executive Officers, if a change of control occurs, a three-year employment period would go into effect. During the employment period, the executive would be entitled to:

receive the highest monthly salary the executive received at any time during the 12-month period preceding the change in control;

receive an annual incentive payment equal to the higher of the executive starget Performance-Based Pay plan incentive or the average of his annual incentive payments for the three years preceding the year in which the change in control occurs;

continue to accrue age and service credit under our qualified and non-qualified defined benefit retirement plans; and

participate in fringe benefit programs that are at least as favorable as those in which the executive was participating prior to the change in control.

If the executive s employment is terminated by the Company without cause or by the executive for good reason during the employment period (or, in certain circumstances, if such a termination occurs prior to and in connection with a change in control), the executive would be entitled to receive a lump-sum payment equal to the value of the payments and benefits identified above that the executive would have received had he continued to be employed for the entire employment period. (The terms cause, good reason and change in control are each defined in the change in control agreements.) In the event that the executive s benefits under the agreement are subject to the excise tax imposed under Section 280G of the Internal Revenue Code, the Company will make a tax payment to the executive so that the net amount of such payment (after taxes) he receives is sufficient to pay the excise tax due.

In addition, outstanding and unvested stock options, restricted stock units and the target number of performance stock units would become vested under the terms of our equity plans. In the case of awards granted under the 2004 Long-Term Incentive Equity Plan, unless the Board determined otherwise, the awards would vest upon a change in control irrespective of a termination of employment. Under the 2008 Performance Incentive Plan, awards will not vest unless a termination of employment without cause or for good reason also occurs or an acquirer does not assume outstanding awards. Finally, the executive s unvested benefits under the Supplementary Retirement Plan would vest on a change in control whether or not the executive s employment was terminated. The outstanding equity awards held by the executives as of December 31, 2010 are described above under Outstanding Equity Awards at Fiscal Year End and each executive s accrued benefits under our retirement plans are described above under 2010 Pension Benefits.

In addition, in the event the executive semployment terminates by reason of death, disability or retirement, (i) restricted stock units would become vested under the terms of our equity plans, and performance stock units would vest at the conclusion of the performance period based on actual performance and a proration representing the portion of the performance period employed; and (ii) options would be fully vested upon death or disability, vested to the extent they would have vested in the next three years upon retirement, and the options can be exercised for three years following term of employment. Furthermore, Mr. Ayer would be entitled to lifetime air travel on Alaska Airlines and Horizon Air having an approximate incremental cost to the Company of \$10,956.

67

In the tables below, we have estimated the potential cost to us of the payments and benefits each Named Executive Officer would have received if his employment had terminated due to retirement, death or disability, or change in control on December 31, 2010. As described above, except for the equity acceleration value, the amount an executive would be entitled to receive would be reduced pro-rata for any period the executive actually worked during the employment period.

# Retirement

	Ca Sever	sh rance	Retire	nced ement efit	nefit nuation	Air	time fare fit(4)	Acc	Equity eleration(5)	e Tax s-Up	Total
William S. Ayer	\$	0	\$	0	\$ 0	\$ 10	,956	\$	11,229,094	\$ 0	\$ 11,240,050
Glenn S. Johnson	\$	0	\$	0	\$ 0	\$	0	\$	3,263,348	\$ 0	\$ 3,263,348
Bradley D. Tilden	\$	0	\$	0	\$ 0	\$	0	\$	4,472,681	\$ 0	\$ 4,472,681
Benito Minicucci	\$	0	\$	0	\$ 0	\$	0	\$	3,897,357	\$ 0	\$ 3,897,357
Jeffrey D. Pinneo	\$	0	\$	0	\$ 0	\$	0	\$	2,395,380	\$ 0	\$ 2,395,380
Brandon S. Johnson	\$	0	\$	0	\$ 0	\$	0	\$	1,216,555	\$ 0	\$ 1,216,555

**Death or Disability** 

	Enhanced						Life	time					
	Ca	sh	Retire	ement	Bei	nefit	Air	fare		Equity	Excis	e Tax	
	Sever	rance	Ben	Benefit Continuation		nuation	Benefit(4)		Acceleration(5)		Gross-Up		Total
William S. Ayer	\$	0	\$	0	\$	0	\$ 10	),956	\$	11,455,210	\$	0	\$ 11,466,266
Glenn S. Johnson	\$	0	\$	0	\$	0	\$	0	\$	3,320,751	\$	0	\$ 3,320,751
Bradley D. Tilden	\$	0	\$	0	\$	0	\$	0	\$	4,561,715	\$	0	\$ 4,561,715
Benito Minicucci	\$	0	\$	0	\$	0	\$	0	\$	3,951,832	\$	0	\$ 3,951,832
Jeffrey D. Pinneo	\$	0	\$	0	\$	0	\$	0	\$	2,441,068	\$	0	\$ 2,441,068
Brandon S. Johnson	\$	0	\$	0	\$	0	\$	0	\$	1,231,550	\$	0	\$ 1,231,550

**Change in Control** 

	Cash Severance(1)	Enhanced Retirement Benefit(2)	Benefit tinuation(3)	Air	etime fare efit(4)	Ac	Equity celeration(5)	Excise Tax Gross-Up	Total
William S. Ayer	\$ 2,752,539	\$ 181,460	\$ 165,951	\$ 10	0,956	\$	11,878,596	\$ 1,960,310	\$ 16,949,812
Glenn S. Johnson	\$ 1,815,589	\$ 764,464	\$ 128,054	\$	0	\$	3,532,394	\$ 1,273,432	\$ 7,513,932
Bradley D. Tilden	\$ 2,196,610	\$ 457,923	\$ 153,020	\$	0	\$	4,890,517	\$ 1,564,495	\$ 9,262,585
Benito Minicucci	\$ 1,648,049	\$ 14,700	\$ 118,002	\$	0	\$	4,152,136	\$ 1,141,980	\$ 7,074,796
Jeffrey D. Pinneo	\$ 1,260,000	\$ 431,339	\$ 99,560	\$	0	\$	2,611,138	\$ 822,806	\$ 5224,843
Brandon S. Johnson	\$ 1,178,694	\$ 13,163	\$ 96,048	\$	0	\$	1,287,106	\$ 570,250	\$ 3,145,262

- (1) Represents the amount obtained by multiplying three by the sum of the executive s highest rate of base salary during the preceding 12 months and the higher of the executive s target incentive or his average incentive for the three preceding years.
- (2) Represents the sum of (a) except in the case of Mr. Minicucci and Mr. Pedersen, the actuarial equivalent of an additional three years of age and service credit under our qualified and non-qualified retirement plan using the executive s highest rate of salary during the preceding 12-months prior to a change in control, (b) except in the case of Mr. Minicucci and Mr. Pedersen, the present value of the accrued but unvested portion of the non-qualified retirement benefits that would vest upon a change of control, (c) the matching contribution the

executive would have received under our qualified defined contribution plan had the executive continued to contribute the maximum allowable amount during the employment period, and (d) in the case of Mr. Minicucci and Mr. Pedersen, the contribution the executive would have received under our nonqualified defined contribution plan had the executive continued to participate in the plan during the employment period.

- (3) Represents the estimated cost of (a) 18 months of premiums under our medical, dental and vision programs, and (b) three years of continued participation in life, disability, accidental death insurance and other fringe benefit programs.
- (4) Mr. Ayer is entitled to lifetime air travel benefits under all termination scenarios. In this column, we show the present value of this benefit, calculated using a discount rate and mortality table that are the same as those used for our pension plan accounting under ASC 715-20 as of December 31, 2010, described above in the section titled 2010 Pension Benefits. Other assumptions include that the lifetime average annual usage is equal to actual average annual usage amounts in 2008 through 2010, and that the annual value of the benefit is equal to the annual incremental cost to the Company, which will be the same as the average of the incremental cost incurred to provide air benefits to the executive in those years as disclosed under All Other Compensation in the Summary Compensation Table.
- (5) Represents the in-the-money value of unvested stock options and the face value of unvested restricted stock and performance stock unit awards that would vest upon termination of employment in the circumstances described above based on a stock price of \$56.69 (the closing price of our stock on the last trading day of fiscal 2010). The value of the extended term of the options is not reflected in the table because we have assumed that the executive s outstanding stock options would be assumed by the acquiring company pursuant to a change in control. This calculation is an estimate for proxy disclosure purposes only. Payments on an actual change in control or termination may differ based on factors such as transaction price, timing of employment termination and payments, methodology for valuing stock options, changes in compensation, and reasonable compensation analyses.

# REDUCE DUPLICATIVE MAILINGS

The Company is required to provide an annual report and proxy statement to all stockholders of record. If you have more than one account in your name or at the same address as other stockholders, the Company or your broker may discontinue mailings of multiple copies. If you wish to receive separate mailings for multiple accounts at the same address, you should mark the designated box on your proxy card. If you are voting by telephone or the internet and you wish to receive multiple copies, you may notify us at the address and phone number at the end of the following paragraph if you are a stockholder of record or notify your broker if you hold through a broker.

Once you have received notice from your broker or us that they or we will discontinue sending multiple copies to the same address, you will receive only one copy until you are notified otherwise or until you revoke your consent. If, at any time, you wish to resume receiving separate proxy statements or annual reports, or if you are receiving multiple statements and reports and wish to receive only one, please notify your broker if your shares are held in a brokerage account or us if you hold registered shares. You can notify us by sending a written request to the Company s Corporate Secretary, Alaska Air Group, Inc., P.O. Box 68947, Seattle, WA 98168, or by calling (206) 392-5131.

69

# SUBMISSION OF PROPOSALS FOR NEXT ANNUAL MEETING

The Company expects to hold its next annual meeting on or about May 15, 2012. If you wish to submit a proposal for inclusion in the proxy materials for that meeting, you must send the proposal to the Corporate Secretary at the address below. The proposal must be received at the Company s executive offices no later than December 6, 2011, to be considered for inclusion. Among other requirements set forth in the SEC s proxy rules and the Company s Bylaws, you must have continuously held at least \$2,000 in market value or 1% of the Company s outstanding stock for at least one year by the date of submitting the proposal, and you must continue to own such stock through the date of the meeting.

If you intend to nominate candidates for election as directors or present a proposal at the meeting without including it in the Company s proxy materials, you must provide notice of such proposal to the Company no later than January 18, 2012. The Company s Bylaws outline procedures for giving the required notice. If you would like a copy of the procedures contained in our Bylaws, please contact:

Corporate Secretary

Alaska Air Group, Inc.

P.O. Box 68947

Seattle, WA 98168

70

**EXHIBIT A** 

# AMENDED 2008 ALASKA AIR GROUP, INC. PERFORMANCE INCENTIVE PLAN ALASKA AIR GROUP, INC.

#### 2008 PERFORMANCE INCENTIVE PLAN

# 1. PURPOSE OF PLAN

The purpose of this Alaska Air Group, Inc. 2008 Performance Incentive Plan (this **Plan**) of Alaska Air Group, Inc., a Delaware corporation (the **Company**), is to promote the success of the Company and to increase stockholder value by providing an additional means through the grant of awards to attract, motivate, retain and reward selected employees and other eligible persons.

#### 2. ELIGIBILITY

The Administrator (as such term is defined in Section 3.1) may grant awards under this Plan only to those persons that the Administrator determines to be Eligible Persons. An Eligible Person is any person who is either: (a) an officer (whether or not a director) or employee of the Company or one of its Subsidiaries; (b) a director of the Company or one of its Subsidiaries; or (c) an individual consultant or advisor who renders or has rendered bona fide services (other than services in connection with the offering or sale of securities of the Company or one of its Subsidiaries) to the Company or one of its Subsidiaries and who is selected to participate in this Plan by the Administrator; provided, however, that a person who is otherwise an Eligible Person under clause (c) above may participate in this Plan only if such participation would not adversely affect either the Company s eligibility to use Form S-8 to register under the Securities Act of 1933, as amended (the Securities Act), the offering and sale of shares issuable under this Plan by the Company or the Company s compliance with any other applicable laws. An Eligible Person who has been granted an award (a participant) may, if otherwise eligible, be granted additional awards if the Administrator shall so determine. As used herein, Subsidiary means any corporation or other entity a majority of whose outstanding voting stock or voting power is beneficially owned directly or indirectly by the Company; and Board means the Board of Directors of the Company.

# 3. PLAN ADMINISTRATION

3.1 The Administrator. This Plan shall be administered by and all awards under this Plan shall be authorized by the Administrator. The Administrator means the Board or one or more committees appointed by the Board or another committee (within its delegated authority) to administer all or certain aspects of this Plan. Any such committee shall be comprised solely of one or more directors or such number of directors as may be required under applicable law. A committee may delegate some or all of its authority to another committee so constituted. The Board or a committee comprised solely of directors may also delegate, to the extent permitted by Section 157(c) of the Delaware General Corporation Law and any other applicable law, to one or more officers of the Company, its powers under this Plan (a) to designate the officers and employees of the Company and its Subsidiaries who will

A-1

receive grants of awards under this Plan, and (b) to determine the number of shares subject to, and the other terms and conditions of, such awards. The Board may delegate different levels of authority to different committees with administrative and grant authority under this Plan. Unless otherwise provided in the Bylaws of the Company or the applicable charter of any Administrator: (a) a majority of the members of the acting Administrator shall constitute a quorum, and (b) the vote of a majority of the members present assuming the presence of a quorum or the unanimous written consent of the members of the Administrator shall constitute action by the acting Administrator.

With respect to awards intended to satisfy the requirements for performance-based compensation under Section 162(m) of the Internal Revenue Code of 1986, as amended (the **Code**), this Plan shall be administered by a committee consisting solely of two or more outside directors (as this requirement is applied under Section 162(m) of the Code); provided, however, that the failure to satisfy such requirement shall not affect the validity of the action of any committee otherwise duly authorized and acting in the matter. Award grants, and transactions in or involving awards, intended to be exempt under Rule 16b-3 under the Securities Exchange Act of 1934, as amended (the **Exchange Act**), must be duly and timely authorized by the Board or a committee consisting solely of two or more non-employee directors (as this requirement is applied under Rule 16b-3 promulgated under the Exchange Act). To the extent required by any applicable listing agency, this Plan shall be administered by a committee composed entirely of independent directors (within the meaning of the applicable listing agency).

- **3.2** *Powers of the Administrator*. Subject to the express provisions of this Plan, the Administrator is authorized and empowered to do all things necessary or desirable in connection with the authorization of awards and the administration of this Plan (in the case of a committee or delegation to one or more officers, within the authority delegated to that committee or person(s)), including, without limitation, the authority to:
  - (a) determine eligibility and, from among those persons determined to be eligible, the particular Eligible Persons who will receive an award under this Plan;
  - (b) grant awards to Eligible Persons, determine the price at which securities will be offered or awarded and the number of securities to be offered or awarded to any of such persons, determine the other specific terms and conditions of such awards consistent with the express limits of this Plan, establish the installments (if any) in which such awards shall become exercisable or shall vest (which may include, without limitation, performance and/or time-based schedules), or determine that no delayed exercisability or vesting is required, establish any applicable performance targets, and establish the events of termination or reversion of such awards;
  - (c) approve the forms of award agreements (which need not be identical either as to type of award or among participants);
  - (d) construe and interpret this Plan and any agreements defining the rights and obligations of the Company, its Subsidiaries, and participants under this Plan, further define the terms used in this Plan, and prescribe, amend and rescind rules and regulations relating to the administration of this Plan or the awards granted under this Plan;

A-2

- (e) cancel, modify, or waive the Company s rights with respect to, or modify, discontinue, suspend, or terminate any or all outstanding awards, subject to any required consent under Section 8.6.5;
- (f) accelerate or extend the vesting or exercisability or extend the term of any or all such outstanding awards (in the case of options or stock appreciation rights, within the maximum ten-year term of such awards) in such circumstances as the Administrator may deem appropriate (including, without limitation, in connection with a termination of employment or services or other events of a personal nature) subject to any required consent under Section 8.6.5;
- (g) adjust the number of shares of Common Stock subject to any award, adjust the price of any or all outstanding awards or otherwise change previously imposed terms and conditions, in such circumstances as the Administrator may deem appropriate, in each case subject to Sections 4 and 8.6 (and subject to the no repricing provision below);
- (h) determine the date of grant of an award, which may be a designated date after but not before the date of the Administrator s action (unless otherwise designated by the Administrator, the date of grant of an award shall be the date upon which the Administrator took the action granting an award);
- (i) determine whether, and the extent to which, adjustments are required pursuant to Section 7 hereof and authorize the termination, conversion, substitution or succession of awards upon the occurrence of an event of the type described in Section 7;
- (j) acquire or settle (subject to Sections 7 and 8.6) rights under awards in cash, stock of equivalent value, or other consideration (subject to the no repricing provision below); and
- (k) determine the fair market value of the Common Stock or awards under this Plan from time to time and/or the manner in which such value will be determined.

Notwithstanding the foregoing and except for an adjustment pursuant to Section 7.1 or a repricing approved by stockholders, in no case may the Administrator (1) amend an outstanding stock option or SAR to reduce the exercise price or base price of the award, (2) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for cash or other awards for the purpose of repricing the award, or (3) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for an option or SAR with an exercise or base price that is less than the exercise or base price of the original award.

3.3 Binding Determinations. Any action taken by, or inaction of, the Company, any Subsidiary, or the Administrator relating or pursuant to this Plan and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. Neither the Board nor any Board committee, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan (or any award made under this Plan), and all such persons shall be entitled to indemnification and reimbursement by the Company in respect of any claim, loss, damage or expense

A-3

(including, without limitation, attorneys fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time.

- **3.4** Reliance on Experts. In making any determination or in taking or not taking any action under this Plan, the Administrator may obtain and may rely upon the advice of experts, including employees and professional advisors to the Company. No director, officer or agent of the Company or any of its Subsidiaries shall be liable for any such action or determination taken or made or omitted in good faith.
- **3.5 Delegation**. The Administrator may delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Company or any of its Subsidiaries or to third parties.

# 4. SHARES OF COMMON STOCK SUBJECT TO THE PLAN; SHARE LIMITS

- **4.1 Shares Available.** Subject to the provisions of Section 7.1, the capital stock that may be delivered under this Plan shall be shares of the Company s authorized but unissued Common Stock and any shares of its Common Stock held as treasury shares. For purposes of this Plan, **Common Stock** shall mean the common stock of the Company and such other securities or property as may become the subject of awards under this Plan, or may become subject to such awards, pursuant to an adjustment made under Section 7.1.
- **4.2 Share Limits.** The maximum number of shares of Common Stock that may be delivered pursuant to awards granted to Eligible Persons under this Plan (the **Share Limit**) is equal to the sum of the following:
  - (1) 4,300,000 shares of Common Stock, plus
  - (2) the number of shares of Common Stock available for additional award grant purposes under the Company s 2004 Long-Term Incentive Plan (the **2004 Plan**) as of the date of stockholder approval of this Plan (the **Stockholder Approval Date**) and determined immediately prior to the termination of the authority to grant new awards under the 2004 Plan as of the Stockholder Approval Date, plus
  - (3) the number of any shares subject to stock options granted under the 2004 Plan and outstanding on the Stockholder Approval Date which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date without being exercised (including any shares subject to stock options granted under the Company s 1999 Long-Term Equity Incentive Plan (the 1999 Plan) outstanding on the Stockholder Approval Date which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date and would otherwise become eligible for award grant purposes under the 2004 Plan in accordance with the provisions of that plan);
  - (4) the number of any shares subject to restricted stock and restricted stock unit awards granted under the 2004 Plan that are outstanding and unvested on the Stockholder Approval Date that are forfeited, terminated, cancelled or otherwise reacquired by the Company without having become vested.

provided that in no event shall the Share Limit exceed 7,299,517 shares (which is the sum of the 4,300,000 shares set forth above, plus the number of shares

A-4

available under the 2004 Plan for additional award grant purposes as of the Effective Date (as such term is defined in Section 8.6.1), plus the aggregate number of shares subject to awards previously granted and outstanding under the 2004 Plan and the 1999 Plan as of the Effective Date).

Shares issued in respect of any Full-Value Award granted under this Plan shall be counted against the foregoing Share Limit as [1.7] shares for every one share actually issued in connection with such award. (For example, if a stock bonus of 100 shares of Common Stock is granted under this Plan, [170] shares shall be charged against the Share Limit in connection with that award.) For this purpose, a **Full-Value Award** means any award under this Plan that is <u>not</u> a stock option grant or a stock appreciation right grant.

The following limits also apply with respect to awards granted under this Plan:

- (a) The maximum number of shares of Common Stock that may be delivered pursuant to options qualified as incentive stock options granted under this Plan is 4,300,000 shares.
- (b) The maximum number of shares of Common Stock subject to those options and stock appreciation rights that are granted during any calendar year to any individual under this Plan is 300,000 shares.
- (c) Additional limits with respect to Performance-Based Awards are set forth in Section 5.2.3. Each of the foregoing numerical limits is subject to adjustment as contemplated by Section 4.3, Section 7.1, and Section 8.10.
  - 4.3 Awards Settled in Cash, Reissue of Awards and Shares. Except as provided in the next sentence, shares that are subject to or underlie awards granted under this Plan which expire or for any reason are cancelled or terminated, are forfeited, fail to vest, or for any other reason are not paid or delivered under this Plan shall again be available for subsequent awards under this Plan. Shares that are exchanged by a participant or withheld by the Company as full or partial payment in connection with any award under this Plan, as well as any shares exchanged by a participant or withheld by the Company or one of its Subsidiaries to satisfy the tax withholding obligations related to any award, shall not be available for subsequent awards under this Plan. To the extent that an award granted under this Plan is settled in cash or a form other than shares of Common Stock, the shares that would have been delivered had there been no such cash or other settlement shall not be counted against the shares available for issuance under this Plan. In the event that shares of Common Stock are delivered in respect of a dividend equivalent right granted under this Plan, the number of shares delivered with respect to the award shall be counted against the share limits of this Plan (including, for purposes of clarity, the limits of Section 4.2 of this Plan). (For purposes of clarity, if 1,000 dividend equivalent rights are granted and outstanding when the Company pays a dividend, and 50 shares are delivered in payment of those rights with respect to that dividend, 85 shares (after giving effect to the Full-Value Award premium counting rules) shall be counted against the share limits of this Plan). To the extent that shares of Common Stock are delivered pursuant to the exercise of a stock appreciation right or stock option granted under this Plan, the number of underlying shares as to which the exercise

A-5

related shall be counted against the applicable share limits under Section 4.2, as opposed to only counting the shares-issued. (For purposes of clarity, if a stock appreciation right relates to 100,000 shares and is exercised at a time when the payment due to the participant is 15,000 shares, 100,000 shares shall be charged against the applicable share limits under Section 4.2 with respect to such exercise). Refer to Section 8.10 for application of the foregoing share limits with respect to assumed awards. The foregoing adjustments to the share limits of this Plan are subject to any applicable limitations under Section 162(m) of the Code with respect to awards intended as performance-based compensation thereunder.

**4.4** Reservation of Shares; No Fractional Shares; Minimum Issue. The Company shall at all times reserve a number of shares of Common Stock sufficient to cover the Company sobligations and contingent obligations to deliver shares with respect to awards then outstanding under this Plan (exclusive of any dividend equivalent obligations to the extent the Company has the right to settle such rights in cash). No fractional shares shall be delivered under this Plan. The Administrator may pay cash in lieu of any fractional shares in settlements of awards under this Plan. The Administrator may from time to time impose a limit (of not greater than 100 shares) on the minimum number of shares that may be purchased or exercised as to awards granted under this Plan unless (as to any particular award) the total number purchased or exercised is the total number at the time available for purchase or exercise under the award.

#### 5. AWARDS

- **5.1** *Type and Form of Awards*. The Administrator shall determine the type or types of award(s) to be made to each selected Eligible Person. Awards may be granted singly, in combination or in tandem. Awards also may be made in combination or in tandem with, in replacement of, as alternatives to, or as the payment form for grants or rights under any other employee or compensation plan of the Company or one of its Subsidiaries. The types of awards that may be granted under this Plan are (subject, in each case, to the no repricing provisions of Section 3.2):
  - 5.1.1 Stock Options. A stock option is the grant of a right to purchase a specified number of shares of Common Stock during a specified period as determined by the Administrator. An option may be intended as an incentive stock option within the meaning of Section 422 of the Code (an ISO) or a nonqualified stock option (an option not intended to be an ISO). The award agreement for an option will indicate if the option is intended as an ISO; otherwise it will be deemed to be a nonqualified stock option. The maximum term of each option (ISO or nonqualified) shall be ten (10) years. The per share exercise price for each option shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the option. When an option is exercised, the exercise price for the shares to be purchased shall be paid in full in cash or such other method permitted by the Administrator consistent with Section 5.5.
  - **5.1.2 Additional Rules Applicable to ISOs.** To the extent that the aggregate fair market value (determined at the time of grant of the applicable option) of stock with respect to which ISOs first become exercisable by a participant in any

A-6

calendar year exceeds \$100,000, taking into account both Common Stock subject to ISOs under this Plan and stock subject to ISOs under all other plans of the Company or one of its Subsidiaries (or any parent or predecessor corporation to the extent required by and within the meaning of Section 422 of the Code and the regulations promulgated thereunder), such options shall be treated as nonqualified stock options. In reducing the number of options treated as ISOs to meet the \$100,000 limit, the most recently granted options shall be reduced first. To the extent a reduction of simultaneously granted options is necessary to meet the \$100,000 limit, the Administrator may, in the manner and to the extent permitted by law, designate which shares of Common Stock are to be treated as shares acquired pursuant to the exercise of an ISO. ISOs may only be granted to employees of the Company or one of its subsidiaries (for this purpose, the term subsidiary is used as defined in Section 424(f) of the Code, which generally requires an unbroken chain of ownership of at least 50% of the total combined voting power of all classes of stock of each subsidiary in the chain beginning with the Company and ending with the subsidiary in question). There shall be imposed in any award agreement relating to ISOs such other terms and conditions as from time to time are required in order that the option be an incentive stock option as that term is defined in Section 422 of the Code. No ISO may be granted to any person who, at the time the option is granted, owns (or is deemed to own under Section 424(d) of the Code) shares of outstanding Common Stock possessing more than 10% of the total combined voting power of all classes of stock of the Company, unless the exercise price of such option is at least 110% of the fair market value of the stock subject to the option and such option by its terms is not exercisable after the expiration of five years from the date such option is granted.

- 5.1.3 Stock Appreciation Rights. A stock appreciation right or SAR is a right to receive a payment, in cash and/or Common Stock, equal to the excess of the fair market value of a specified number of shares of Common Stock on the date the SAR is exercised over the base price of the award, which base price shall be set forth in the applicable award agreement and shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the SAR. The maximum term of a SAR shall be ten (10) years.
- 5.1.4 Other Awards; Dividend Equivalent Rights. The other types of awards that may be granted under this Plan include:

  (a) stock bonuses, restricted stock, performance stock, stock units, phantom stock or similar rights to purchase or acquire shares, whether at a fixed or variable price or ratio related to the Common Stock, upon the passage of time, the occurrence of one or more events, or the satisfaction of performance criteria or other conditions, or any combination thereof; (b) any similar securities with a value derived from the value of or related to the Common Stock and/or returns thereon; or (c) cash awards. Dividend equivalent rights may be granted as a separate award or in connection with another award under this Plan; provided, however, that dividend equivalent rights may not be granted in connection with a stock option or SAR granted under this Plan. In addition, any dividends and/or dividend equivalents as to the unvested portion of a restricted stock award

A-7

that is subject to performance-based vesting requirements or the unvested portion of a stock unit award that is subject to performance-based vesting requirements will be subject to termination and forfeiture to the same extent as the corresponding portion of the award to which they relate.

- 5.2 Section 162(m) Performance-Based Awards. Without limiting the generality of the foregoing, any of the types of awards listed in Section 5.1.4 above may be, and options and SARs granted to officers and employees ( Qualifying Options and Qualifying SARS, respectively) typically will be, granted as awards intended to satisfy the requirements for performance-based compensation within the meaning of Section 162(m) of the Code ( Performance-Based Awards ). The grant, vesting, exercisability or payment of Performance-Based Awards may depend (or, in the case of Qualifying Options or Qualifying SARs, may also depend) on the degree of achievement of one or more performance goals relative to a pre-established targeted level or levels using one or more of the Business Criteria set forth below (on an absolute basis or relative to the performance of other companies or upon comparisons of any of the indicators of performance relative to other companies) for the Company on a consolidated basis or for one or more of the Company s subsidiaries, segments, divisions or business units, or any combination of the foregoing. Any Qualifying Option or Qualifying SAR shall be subject only to the requirements of Section 5.2.1 and 5.2.3 in order for such award to satisfy the requirements for performance-based compensation under Section 162(m) of the Code. Any other Performance-Based Award shall be subject to all of the following provisions of this Section 5.2.
  - **5.2.1** Class; Administrator. The eligible class of persons for Performance-Based Awards under this Section 5.2 shall be officers and employees of the Company or one of its Subsidiaries. The Administrator approving Performance-Based Awards or making any certification required pursuant to Section 5.2.4 must be constituted as provided in Section 3.1 for awards that are intended as performance-based compensation under Section 162(m) of the Code.
  - 5.2.2 Performance Goals. The specific performance goals for Performance-Based Awards (other than Qualifying Options and Qualifying SARs) shall be, on an absolute or relative basis, established based on one or more of the following business criteria (Business Criteria) as selected by the Administrator in its sole discretion: earnings per share, cash flow (which means cash and cash equivalents derived from either net cash flow from operations or net cash flow from operations, financing and investing activities), stock price, total stockholder return, gross revenue, revenue growth, operating income (before or after taxes), net earnings (before or after interest, taxes, depreciation and/or amortization), return on equity or on assets or on net investment, cost containment or reduction, profitability, economic value added, market share, productivity, safety, customer satisfaction, on-time performance, or any combination thereof. These terms are used as applied under generally accepted accounting principles or in the financial reporting of the Company or of its Subsidiaries. To qualify awards as performance-based under Section 162(m), the applicable Business Criterion (or Business Criteria, as the case may be) and specific performance goal or goals (targets) must be established and approved by the Administrator during the first 90 days of the

A-8

performance period (and, in the case of performance periods of less than one year, in no event after 25% or more of the performance period has elapsed) and while performance relating to such target(s) remains substantially uncertain within the meaning of Section 162(m) of the Code. The terms of the Performance-Based Awards may specify the manner, if any, in which performance targets shall be adjusted to mitigate the unbudgeted impact of material, unusual or nonrecurring gains and losses, accounting changes or other extraordinary events not foreseen at the time the targets were set unless the Administrator provides otherwise at the time of establishing the targets. The applicable performance measurement period may not be less than three months nor more than 10 years.

- 5.2.3 Form of Payment; Maximum Performance-Based Award. Grants or awards under this Section 5.2 may be paid in cash or shares of Common Stock or any combination thereof. Grants of Qualifying Options and Qualifying SARs to any one participant in any one calendar year shall be subject to the limit set forth in Section 4.2(b). The maximum number of shares of Common Stock which may be delivered pursuant to Performance-Based Awards (other than Qualifying Options and Qualifying SARs, and other than cash awards covered by the following sentence) that are granted to any one participant in any one calendar year shall not exceed 300,000 shares, either individually or in the aggregate, subject to adjustment as provided in Section 7.1. In addition, the aggregate amount of compensation to be paid to any one participant in respect of all Performance-Based Awards payable only in cash and not related to shares of Common Stock and granted to that participant in any one calendar year shall not exceed \$1,000,000. Awards that are cancelled during the year shall be counted against these limits to the extent required by Section 162(m) of the Code.
- **5.2.4** *Certification of Payment.* Before any Performance-Based Award under this Section 5.2 (other than Qualifying Options and Qualifying SARs) is paid and to the extent required to qualify the award as performance-based compensation within the meaning of Section 162(m) of the Code, the Administrator must certify in writing that the performance target(s) and any other material terms of the Performance-Based Award were in fact timely satisfied.
- **5.2.5** Reservation of Discretion. The Administrator will have the discretion to determine the restrictions or other limitations of the individual awards granted under this Section 5.2 including the authority to reduce awards, payouts or vesting or to pay no awards, in its sole discretion, if the Administrator preserves such authority at the time of grant by language to this effect in its authorizing resolutions or otherwise.
- **5.2.6** Expiration of Grant Authority. As required pursuant to Section 162(m) of the Code and the regulations promulgated thereunder, the Administrator s authority to grant new awards that are intended to qualify as performance-based compensation within the meaning of Section 162(m) of the Code (other than Qualifying Options and Qualifying SARs) shall terminate upon the first meeting of the Company s stockholders that occurs in the fifth year following the year in which the Company s stockholders first approve this Plan, subject to any subsequent extension that may be approved by stockholders.

A-9

- 5.3 Award Agreements. Each award shall be evidenced by either (1) a written award agreement in a form approved by the Administrator and executed by the Company by an officer duly authorized to act on its behalf, or (2) an electronic notice of award grant in a form approved by the Administrator and recorded by the Company (or its designee) in an electronic recordkeeping system used for the purpose of tracking award grants under this Plan generally (in each case, an award agreement), as the Administrator may provide and, in each case and if required by the Administrator, executed or otherwise electronically accepted by the recipient of the award in such form and manner as the Administrator may require. The Administrator may authorize any officer of the Company (other than the particular award recipient) to execute any or all award agreements on behalf of the Company. The award agreement shall set forth the material terms and conditions of the award as established by the Administrator consistent with the express limitations of this Plan.
- 5.4 Deferrals and Settlements. Payment of awards may be in the form of cash, Common Stock, other awards or combinations thereof as the Administrator shall determine, and with such restrictions as it may impose. The Administrator may also require or permit participants to elect to defer the issuance of shares or the settlement of awards in cash under such rules and procedures as it may establish under this Plan. The Administrator may also provide that deferred settlements include the payment or crediting of interest or other earnings on the deferral amounts, or the payment or crediting of dividend equivalents where the deferred amounts are denominated in shares.
- **5.5** Consideration for Common Stock or Awards. The purchase price for any award granted under this Plan or the Common Stock to be delivered pursuant to an award, as applicable, may be paid by means of any lawful consideration as determined by the Administrator, including, without limitation, one or a combination of the following methods:

services rendered by the recipient of such award;

cash, check payable to the order of the Company, or electronic funds transfer;

notice and third party payment in such manner as may be authorized by the Administrator;

the delivery of previously owned shares of Common Stock;

by a reduction in the number of shares otherwise deliverable pursuant to the award; or

subject to such procedures as the Administrator may adopt, pursuant to a cashless exercise with a third party who provides financing for the purposes of (or who otherwise facilitates) the purchase or exercise of awards.

In no event shall any shares newly-issued by the Company be issued for less than the minimum lawful consideration for such shares or for consideration other than consideration permitted by applicable state law. Shares of Common Stock used to satisfy the exercise price of an option shall be valued at their fair market value on the date of exercise. The Company will not be obligated to deliver any shares unless and until it receives full payment of the exercise or purchase price therefor and any related withholding obligations under Section 8.5 and any other conditions to exercise or purchase have been satisfied. Unless otherwise expressly provided in the

A-10

applicable award agreement, the Administrator may at any time eliminate or limit a participant s ability to pay the purchase or exercise price of any award or shares by any method other than cash payment to the Company.

by the Administrator in the circumstances, the closing price (in regular trading) for a share of Common Stock on the New York Stock Exchange (the Exchange) for the date in question or, if no sales of Common Stock were reported on the Exchange on that date, the closing price (in regular trading) for a share of Common Stock on the Exchange for the next preceding day on which sales of Common Stock were reported on the Exchange. The Administrator may, however, provide with respect to one or more awards that the fair market value shall equal the closing price (in regular trading) for a share of Common Stock on the Exchange on the last trading day preceding the date in question or the average of the high and low trading prices of a share of Common Stock on the Exchange for the date in question or the most recent trading day. If the Common Stock is no longer listed or is no longer actively traded on the Exchange as of the applicable date, the fair market value of the Common Stock shall be the value as reasonably determined by the Administrator for purposes of the award in the circumstances. The Administrator also may adopt a different methodology for determining fair market value with respect to one or more awards if a different methodology is necessary or advisable to secure any intended favorable tax, legal or other treatment for the particular award(s) (for example, and without limitation, the Administrator may provide that fair market value for purposes of one or more awards will be based on an average of closing prices (or the average of high and low daily trading prices) for a specified period preceding the relevant date).

#### 5.7 Transfer Restrictions.

- **5.7.1** *Limitations on Exercise and Transfer*. Unless otherwise expressly provided in (or pursuant to) this Section 5.7 or required by applicable law: (a) all awards are non-transferable and shall not be subject in any manner to sale, transfer, anticipation, alienation, assignment, pledge, encumbrance or charge; (b) awards shall be exercised only by the participant; and (c) amounts payable or shares issuable pursuant to any award shall be delivered only to (or for the account of) the participant.
- **5.7.2** *Exceptions*. The Administrator may permit awards to be exercised by and paid to, or otherwise transferred to, other persons or entities pursuant to such conditions and procedures, including limitations on subsequent transfers, as the Administrator may, in its sole discretion, establish in writing. Any permitted transfer shall be subject to compliance with applicable federal and state securities laws and shall not be for value (other than nominal consideration, settlement of marital property rights, or for interests in an entity in which more than 50% of the voting interests are held by the Eligible Person or by the Eligible Person s family members).
- 5.7.3 Further Exceptions to Limits on Transfer. The exercise and transfer restrictions in Section 5.7.1 shall not apply to:
  - (a) transfers to the Company (for example, in connection with the expiration or termination of the award),

A-11

- (b) the designation of a beneficiary to receive benefits in the event of the participant s death or, if the participant has died, transfers to or exercise by the participant s beneficiary, or, in the absence of a validly designated beneficiary, transfers by will or the laws of descent and distribution,
- (c) subject to any applicable limitations on ISOs, transfers to a family member (or former family member) pursuant to a domestic relations order if approved or ratified by the Administrator,
- (d) if the participant has suffered a disability, permitted transfers or exercises on behalf of the participant by his or her legal representative, or
- (e) the authorization by the Administrator of cashless exercise procedures with third parties who provide financing for the purpose of (or who otherwise facilitate) the exercise of awards consistent with applicable laws and the express authorization of the Administrator.
- **International Awards**. One or more awards may be granted to Eligible Persons who provide services to the Company or one of its Subsidiaries outside of the United States. Any awards granted to such persons may be granted pursuant to the terms and conditions of any applicable sub-plans, if any, appended to this Plan and approved by the Administrator.

#### 6. EFFECT OF TERMINATION OF EMPLOYMENT OR SERVICE ON AWARDS

- 6.1 General. The Administrator shall establish the effect of a termination of employment or service on the rights and benefits under each award under this Plan and in so doing may make distinctions based upon, inter alia, the cause of termination and type of award. If the participant is not an employee of the Company or one of its Subsidiaries and provides other services to the Company or one of its Subsidiaries, the Administrator shall be the sole judge for purposes of this Plan (unless a contract or the award otherwise provides) of whether the participant continues to render services to the Company or one of its Subsidiaries and the date, if any, upon which such services shall be deemed to have terminated.
- 6.2 Events Not Deemed Terminations of Service. Unless the express policy of the Company or one of its Subsidiaries, or the Administrator, otherwise provides, the employment relationship shall not be considered terminated in the case of (a) sick leave, (b) military leave, or (c) any other leave of absence authorized by the Company or one of its Subsidiaries, or the Administrator; provided that, unless reemployment upon the expiration of such leave is guaranteed by contract or law or the Administrator otherwise provides, such leave is for a period of not more than three months. In the case of any employee of the Company or one of its Subsidiaries on an approved leave of absence, continued vesting of the award while on leave from the employ of the Company or one of its Subsidiaries may be suspended until the employee returns to service, unless the Administrator otherwise provides or applicable law otherwise requires. In no event shall an award be exercised after the expiration of the term set forth in the applicable award agreement.
- **6.3** Effect of Change of Subsidiary Status. For purposes of this Plan and any award, if an entity ceases to be a Subsidiary of the Company a termination of employment or service shall be deemed to have occurred with respect to each Eligible Person in

A-12

respect of such Subsidiary who does not continue as an Eligible Person in respect of the Company or another Subsidiary that continues as such after giving effect to the transaction or other event giving rise to the change in status unless the Subsidiary that is sold, spun-off or otherwise divested (or its successor or a direct or indirect parent of such Subsidiary or successor) assumes the Eligible Person s award(s) in connection with such transaction.

#### 7. ADJUSTMENTS; ACCELERATION

7.1 Adjustments. Subject to Section 7.2, upon (or, as may be necessary to effect the adjustment, immediately prior to): any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend) or reverse stock split; any merger, combination, consolidation, or other reorganization; any spin-off, split-up, or similar extraordinary dividend distribution in respect of the Common Stock; or any exchange of Common Stock or other securities of the Company, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; then the Administrator shall equitably and proportionately adjust (1) the number and type of shares of Common Stock (or other securities) that thereafter may be made the subject of awards (including the specific share limits, maximums and numbers of shares set forth elsewhere in this Plan), (2) the number, amount and type of shares of Common Stock (or other securities or property) subject to any outstanding awards, (3) the grant, purchase, or exercise price (which term includes the base price of any SAR or similar right) of any outstanding awards, and/or (4) the securities, cash or other property deliverable upon exercise or payment of any outstanding awards, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding awards.

Unless otherwise expressly provided in the applicable award agreement, upon (or, as may be necessary to effect the adjustment, immediately prior to) any event or transaction described in the preceding paragraph or a sale of all or substantially all of the business or assets of the Company as an entirety, the Administrator shall equitably and proportionately adjust the performance standards applicable to any then-outstanding performance-based awards to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding performance-based awards.

It is intended that, if possible, any adjustments contemplated by the preceding two paragraphs be made in a manner that satisfies applicable U.S. legal, tax (including, without limitation and as applicable in the circumstances, Section 424 of the Code, Section 409A of the Code and Section 162(m) of the Code) and accounting (so as to not trigger any charge to earnings with respect to such adjustment) requirements.

Without limiting the generality of Section 3.3, any good faith determination by the Administrator as to whether an adjustment is required in the circumstances pursuant to this Section 7.1, and the extent and nature of any such adjustment, shall be conclusive and binding on all persons.

**7.2** Corporate Transactions Assumption and Termination of Awards. Upon the occurrence of any of the following: any merger, combination, consolidation, or other reorganization in connection with which the Company does not survive (or does not survive as a public company in respect of its Common Stock); any exchange of

A-13

Common Stock or other securities of the Company in connection with which the Company does not survive (or does not survive as a public company in respect of its Common Stock); a sale of all or substantially all the business, stock or assets of the Company in connection with which the Company does not survive (or does not survive as a public company in respect of its Common Stock); a dissolution of the Company; or any other event in which the Company does not survive (or does not survive as a public company in respect of its Common Stock); then the Administrator may make provision for a cash payment in settlement of, or for the termination, assumption, substitution or exchange of any or all outstanding share-based awards or the cash, securities or property deliverable to the holder of any or all outstanding share-based awards, based upon, to the extent relevant under the circumstances, the distribution or consideration payable to holders of the Common Stock upon or in respect of such event. Upon the occurrence of any event described in the preceding sentence, then, unless the Administrator has made a provision for the substitution, assumption, exchange or other continuation or settlement of the award or (unless the Administrator has provided for the termination of the award) the award would otherwise continue in accordance with its terms in the circumstances: (1) unless otherwise provided in the applicable award agreement, each then-outstanding option and SAR shall become fully vested, all shares of restricted stock then outstanding shall fully vest free of restrictions, and each other award granted under this Plan that is then outstanding shall become payable to the holder of such award; and (2) each award shall terminate upon the related event; provided that the holder of an option or SAR shall be given reasonable advance notice of the impending termination and a reasonable opportunity to exercise his or her outstanding vested options and SARs (after giving effect to any accelerated vesting required in the circumstances) in accordance with their terms before the termination of such awards (except that in no case shall more than ten days notice of the impending termination be required and any acceleration of vesting and any exercise of any portion of an award that is so accelerated may be made contingent upon the actual occurrence of the event).

The Administrator may adopt such valuation methodologies for outstanding awards as it deems reasonable in the event of a cash or property settlement and, in the case of options, SARs or similar rights, but without limitation on other methodologies, may base such settlement solely upon the excess if any of the per share amount payable upon or in respect of such event over the exercise or base price of the award.

In any of the events referred to in this Section 7.2, the Administrator may take such action contemplated by this Section 7.2 prior to such event (as opposed to on the occurrence of such event) to the extent that the Administrator deems the action necessary to permit the participant to realize the benefits intended to be conveyed with respect to the underlying shares. Without limiting the generality of the foregoing, the Administrator may deem an acceleration to occur immediately prior to the applicable event and, in such circumstances, will reinstate the original terms of the award if an event giving rise to an acceleration does not occur.

Without limiting the generality of Section 3.3, any good faith determination by the Administrator pursuant to its authority under this Section 7.2 shall be conclusive and binding on all persons.

A-14

- **7.3** Other Acceleration Rules. The Administrator may override the provisions of Section 7.2 by express provision in the award agreement and may accord any Eligible Person a right to refuse any acceleration, whether pursuant to the award agreement or otherwise, in such circumstances as the Administrator may approve. The portion of any ISO accelerated in connection with an event referred to in Section 7.2 (or such other circumstances as may trigger accelerated vesting of the award) shall remain exercisable as an ISO only to the extent the applicable \$100,000 limitation on ISOs is not exceeded. To the extent exceeded, the accelerated portion of the option shall be exercisable as a nonqualified stock option under the Code.
- 7.4 Definition of Change in Control. With respect to a particular award granted under this Plan, a Change in Control shall be deemed to have occurred as of the first day, after the date of grant of the particular award, that any one or more of the following conditions shall have been satisfied:
  - (i) the consummation of:
    - (A) any consolidation or merger of the Company in which the Company is not the continuing or surviving corporation or pursuant to which shares of common stock of the Company would be converted into cash, securities or other property, other than a merger of the Company in which the holders of common stock of the Company immediately prior to the merger have the same proportionate ownership of common stock of the surviving corporation immediately after the merger; or
    - (B) any sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all, or substantially all, the assets of the Company.
  - (ii) at any time during a period of twenty-four (24) months, fewer than a majority of the members of the Board are Incumbent Directors. For these purposes, **Incumbent Directors** means (A) individuals who constitute the Board at the beginning of such period; and (B) individuals who were nominated or elected by all of, or a committee composed entirely of, the individuals described in (A); and (C) individuals who were nominated or elected by individuals described in (B).
  - (iii) any Person (meaning any individual, entity or group within the meaning of Section 13(d)(3) or 14(d) of the Exchange Act) shall, as a result of a tender or exchange offer, open market purchases, privately-negotiated purchases or otherwise, become the beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act), directly or indirectly, of the then-outstanding securities of the Company ordinarily (and apart from rights accruing under special circumstances) having the right to vote in the election of members of the Board ( Voting Securities to be calculated as provided in paragraph (d) of Rule 13d-3 in the case of rights to acquire common stock of the Company) representing 20% or more of the combined voting power of the then-outstanding Voting Securities.
- (iv) approval by the stockholders of the Company of any plan or proposal for the liquidation or dissolution of the Company. Unless the Board shall determine otherwise, a Change of Control shall not be deemed to have occurred by reason of any corporate reorganization, merger,

A-15

consolidation, transfer of assets, liquidating distribution or other transaction entered into solely by and between the Company and any affiliate thereof, provided such transaction has been approved by at least two-thirds (2/3) of the Incumbent Directors (as defined above) then in office and voting.

#### 8. OTHER PROVISIONS

- 8.1 Compliance with Laws. This Plan, the granting and vesting of awards under this Plan, the offer, issuance and delivery of shares of Common Stock, and/or the payment of money under this Plan or under awards are subject to compliance with all applicable federal and state laws, rules and regulations (including but not limited to state and federal securities law and federal margin requirements) and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Company, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Company or one of its Subsidiaries, provide such assurances and representations to the Company or one of its Subsidiaries as the Administrator may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.
- **8.2** *No Rights to Award*. No person shall have any claim or rights to be granted an award (or additional awards, as the case may be) under this Plan, subject to any express contractual rights (set forth in a document other than this Plan) to the contrary.
- 8.3 No Employment/Service Contract. Nothing contained in this Plan (or in any other documents under this Plan or in any award) shall confer upon any Eligible Person or other participant any right to continue in the employ or other service of the Company or one of its Subsidiaries, constitute any contract or agreement of employment or other service or affect an employee s status as an employee at will, nor shall interfere in any way with the right of the Company or one of its Subsidiaries to change a person s compensation or other benefits, or to terminate his or her employment or other service, with or without cause. Nothing in this Section 8.3, however, is intended to adversely affect any express independent right of such person under a separate employment or service contract other than an award agreement.
- **8.4 Plan Not Funded.** Awards payable under this Plan shall be payable in shares or from the general assets of the Company, and no special or separate reserve, fund or deposit shall be made to assure payment of such awards. No participant, beneficiary or other person shall have any right, title or interest in any fund or in any specific asset (including shares of Common Stock, except as expressly otherwise provided) of the Company or one of its Subsidiaries by reason of any award hereunder. Neither the provisions of this Plan (or of any related documents), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan shall create, or be construed to create, a trust of any kind or a fiduciary relationship between the Company or one of its Subsidiaries and any participant, beneficiary or other person. To the extent that a participant, beneficiary or other person acquires a right to receive payment pursuant to any award hereunder, such right shall be no greater than the right of any unsecured general creditor of the Company.

A-16

- **8.5** *Tax Withholding*. Upon any exercise, vesting, or payment of any award, or upon the disposition of shares of Common Stock acquired pursuant to the exercise of an ISO prior to satisfaction of the holding period requirements of Section 422 of the Code, or upon any other tax withholding event with respect to any award, the Company or one of its Subsidiaries shall have the right at its option to:
  - (a) require the participant (or the participant s personal representative or beneficiary, as the case may be) to pay or provide for payment of at least the minimum amount of any taxes which the Company or one of its Subsidiaries may be required to withhold with respect to such award event or payment; or
  - (b) deduct from any amount otherwise payable in cash (whether related to the award or otherwise) to the participant (or the participant s personal representative or beneficiary, as the case may be) the minimum amount of any taxes which the Company or one of its Subsidiaries may be required to withhold with respect to such award event or payment.

In any case where a tax is required to be withheld in connection with the delivery of shares of Common Stock under this Plan, the Administrator may in its sole discretion (subject to Section 8.1) require or grant (either at the time of the award or thereafter) to the participant the right to elect, pursuant to such rules and subject to such conditions as the Administrator may establish, that the Company reduce the number of shares to be delivered by (or otherwise reacquire) the appropriate number of shares, valued in a consistent manner at their fair market value or at the sales price in accordance with authorized procedures for cashless exercises, necessary to satisfy the minimum applicable withholding obligation on exercise, vesting or payment. In no event shall the shares withheld exceed the minimum whole number of shares required for tax withholding under applicable law.

#### 8.6 Effective Date, Termination and Suspension, Amendments.

**8.6.1** Effective Date. This Plan is effective as of March 13, 2008, the date of its approval by the Board (the Effective Date ). This Plan shall be submitted for and subject to stockholder approval no later than twelve months after the Effective Date. Unless earlier terminated by the Board, this Plan shall terminate at the close of business on the day before the tenth anniversary of the Effective Date. After the termination of this Plan either upon such stated expiration date or its earlier termination by the Board, no additional awards may be granted under this Plan, but previously granted awards (and the authority of the Administrator with respect thereto, including the authority to amend such awards) shall remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of this Plan.

**8.6.2** *Board Authorization*. The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part. No awards may be granted during any period that the Board suspends this Plan.

**8.6.3** *Stockholder Approval.* To the extent then required by applicable law or any applicable listing agency or required under Sections 162, 422 or 424 of the Code to preserve the intended tax consequences of this Plan, or deemed necessary or advisable by the Board, any amendment to this Plan shall be subject to stockholder approval.

A-17

**8.6.4** *Amendments to Awards*. Without limiting any other express authority of the Administrator under (but subject to) the express limits of this Plan, the Administrator by agreement or resolution may waive conditions of or limitations on awards to participants that the Administrator in the prior exercise of its discretion has imposed, without the consent of a participant, and (subject to the requirements of Sections 3.2 and 8.6.5) may make other changes to the terms and conditions of awards. Any amendment or other action that would constitute a repricing of an award is subject to the limitations set forth in Section 3.2.

**8.6.5** *Limitations on Amendments to Plan and Awards*. No amendment, suspension or termination of this Plan or amendment of any outstanding award agreement shall, without written consent of the participant, affect in any manner materially adverse to the participant any rights or benefits of the participant or obligations of the Company under any award granted under this Plan prior to the effective date of such change. Changes, settlements and other actions contemplated by Section 7 shall not be deemed to constitute changes or amendments for purposes of this Section 8.6.

- **8.7** *Privileges of Stock Ownership.* Except as otherwise expressly authorized by the Administrator, a participant shall not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the participant. Except as expressly required by Section 7.1 or otherwise expressly provided by the Administrator, no adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.
- 8.8 Governing Law; Construction; Severability.
- **8.8.1** *Choice of Law*. This Plan, the awards, all documents evidencing awards and all other related documents shall be governed by, and construed in accordance with the laws of the State of Delaware.
- **8.8.2** Severability. If a court of competent jurisdiction holds any provision invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.

#### 8.8.3 Plan Construction.

- (a) Rule 16b-3. It is the intent of the Company that the awards and transactions permitted by awards be interpreted in a manner that, in the case of participants who are or may be subject to Section 16 of the Exchange Act, qualify, to the maximum extent compatible with the express terms of the award, for exemption from matching liability under Rule 16b-3 promulgated under the Exchange Act. Notwithstanding the foregoing, the Company shall have no liability to any participant for Section 16 consequences of awards or events under awards if an award or event does not so qualify.
- (b) <u>Section 162(m)</u>. Awards under Section 5.1.4 to persons described in Section 5.2 that are either granted or become vested, exercisable or payable based on attainment of one or more performance goals related to the Business Criteria, as well as Qualifying Options and Qualifying SARs granted to persons described in Section 5.2, that are approved by a

A-18

committee composed solely of two or more outside directors (as this requirement is applied under Section 162(m) of the Code) shall be deemed to be intended as performance-based compensation within the meaning of Section 162(m) of the Code unless such committee provides otherwise at the time of grant of the award. It is the further intent of the Company that (to the extent the Company or one of its Subsidiaries or awards under this Plan may be or become subject to limitations on deductibility under Section 162(m) of the Code) any such awards and any other Performance-Based Awards under Section 5.2 that are granted to or held by a person subject to Section 162(m) will qualify as performance-based compensation or otherwise be exempt from deductibility limitations under Section 162(m).

- **8.9** Captions. Captions and headings are given to the sections and subsections of this Plan solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Plan or any provision thereof.
- 8.10 Stock-Based Awards in Substitution for Stock Options or Awards Granted by Other Company. Awards may be granted to Eligible Persons in substitution for or in connection with an assumption of employee stock options, SARs, restricted stock or other stock-based awards granted by other entities to persons who are or who will become Eligible Persons in respect of the Company or one of its Subsidiaries, in connection with a distribution, merger or other reorganization by or with the granting entity or an affiliated entity, or the acquisition by the Company or one of its Subsidiaries, directly or indirectly, of all or a substantial part of the stock or assets of the employing entity. The awards so granted need not comply with other specific terms of this Plan, provided the awards reflect only adjustments giving effect to the assumption or substitution consistent with the conversion applicable to the Common Stock in the transaction and any change in the issuer of the security. Any shares that are delivered and any awards that are granted by, or become obligations of, the Company, as a result of the assumption by the Company of, or in substitution for, outstanding awards previously granted by an acquired company (or previously granted by a predecessor employer (or direct or indirect parent thereof) in the case of persons that become employed by the Company or one of its Subsidiaries in connection with a business or asset acquisition or similar transaction) shall not be counted against the Share Limit or other limits on the number of shares available for issuance under this Plan.
- **8.11** *Non-Exclusivity of Plan*. Nothing in this Plan shall limit or be deemed to limit the authority of the Board or the Administrator to grant awards or authorize any other compensation, with or without reference to the Common Stock, under any other plan or authority.
- 8.12 No Corporate Action Restriction. The existence of this Plan, the award agreements and the awards granted hereunder shall not limit, affect or restrict in any way the right or power of the Board or the stockholders of the Company to make or authorize: (a) any adjustment, recapitalization, reorganization or other change in the capital structure or business of the Company or any Subsidiary, (b) any merger, amalgamation, consolidation or change in the ownership of the Company or any

A-19

Subsidiary, (c) any issue of bonds, debentures, capital, preferred or prior preference stock ahead of or affecting the capital stock (or the rights thereof) of the Company or any Subsidiary, (d) any dissolution or liquidation of the Company or any Subsidiary, (e) any sale or transfer of all or any part of the assets or business of the Company or any Subsidiary, or (f) any other corporate act or proceeding by the Company or any Subsidiary. No participant, beneficiary or any other person shall have any claim under any award or award agreement against any member of the Board or the Administrator, or the Company or any employees, officers or agents of the Company or any Subsidiary, as a result of any such action.

- **8.13** Other Company Benefit and Compensation Programs. Payments and other benefits received by a participant under an award made pursuant to this Plan shall not be deemed a part of a participant s compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Company or any Subsidiary, except where the Administrator expressly otherwise provides or authorizes in writing. Awards under this Plan may be made in addition to, in combination with, as alternatives to or in payment of grants, awards or commitments under any other plans or arrangements of the Company or its Subsidiaries.
- **8.14** Clawback Policy. The awards granted under this Plan are subject to the terms of the Company's recoupment, clawback or similar policy as it may be in effect from time to time, as well as any similar provisions of applicable law, any of which could in certain circumstances require repayment or forfeiture of awards or any shares of Common Stock or other cash or property received with respect to the awards (including any value received from a disposition of the shares acquired upon payment of the award

A-20

### **Electronic Voting Instructions**

You can vote by Internet or telephone! Available 24 hours a day, 7 days a week!

Instead of mailing your proxy, you may choose one of the two voting

methods outlined below to vote your proxy.

VALIDATION DETAILS ARE LOCATED BELOW IN THE TITLE

Proxies submitted by the Internet or telephone must be received by 11:59 p.m., Eastern Time, on May 16, 2011.

Vote by Internet

Log on to the Internet and go to

www.envisionreports.com/alk

Follow the steps outlined on the secured website.

Vote by telephone

Call toll free 1-800-652-VOTE (8683) within the USA, US territories & Canada any time on a touch tone telephone. There is NO CHARGE to you for the call.

Follow the instructions provided by the recorded message.

Using a  $\underline{black\ ink}$  pen, mark your votes with an X as shown in this example. Please do not write outside the designated areas.

q IF YOU HAVE NOT VOTED VIA THE INTERNET OR TELEPHONE, FOLD ALONG THE PERFORATION, DETACH AND RETURN q THE BOTTOM PORTION IN THE ENCLOSED ENVELOPE.

## Edgar Filing: RLI CORP - Form 4

# A The Board of Directors recommends a vote <u>FOR</u> all the nominees listed, a vote <u>FOR</u> Proposals 2, 3, and 6, <u>EVERY ONE YEAR</u> for Proposal 4, and a vote <u>AGAINST</u> Proposal 5.

1. Election of Directors:

	For	Against	Abstain		For	Against	Abstain		For	Against	Abstain	
01 - William S. Ayer	••	••	••	02 - Patricia M. Bedient	••	••	••	03 - Marion C. Blakey	••	••	••	
04 - Phyllis J. Campbell		••	••	05 - Jessie J. Knight, Jr.	••	••	••	06 - R. Marc Langland	••	••	••	
07 - Dennis F. Madsen	••			08 - Byron I. Mallott	••			09 - J. Kenneth Thompson	••			
10 - Bradley D. Tilden	••	••	••									

Ratification of Appointment of KPMG LLP as the Company s Independent Accountants.	For 	Against	Abstain	3. Advisory Vote on the Compensation of the Company s Named Executive Officers.	Against	Abstain
4. Advisory Vote on the Frequency of a Vote on Named Executive Officer Compensation.	1 Yr 2 Yr	3 Yrs	Abstain	5. Advisory Vote on Stockholders Right to Act by Written Consent.	Against	Abstain
6. Approval of the Amendment of the Company s 2008 Performance Incentive Plan.	For	Against	Abstain			

# B Authorized Signatures This section must be completed for your vote to be counted. Date and Sign Below

Please sign exactly as name(s) appears hereon. Joint owners should each sign. When signing as attorney, executor, administrator, corporate officer, trustee, guardian, or custodian, please give full title.

Date (mm/dd/yyyy) Please print date below. Signature 1 Please keep signature within the box. Signature 2 Please keep signature within the box.

/ /

 ${\bf q}$   $\,$  IF YOU HAVE NOT VOTED VIA THE INTERNET  $\underline{\bf OR}$  TELEPHONE, FOLD ALONG THE PERFORATION, DETACH AND RETURN THE  $\,$   ${\bf q}$  BOTTOM PORTION IN THE ENCLOSED ENVELOPE.

### PROXY Alaska Air Group, Inc

#### THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS

#### 2011 ANNUAL MEETING OF STOCKHOLDERS

#### MAY 17, 2011

The stockholder hereby appoints William S. Ayer and Keith Loveless, or either of them, as proxies, each with the power to appoint his substitute, and hereby authorizes them to represent and to vote, as designated on the reverse side of this card, all of the shares of Common Stock of Alaska Air Group, Inc. that the stockholder is entitled to vote at the Annual Meeting of Stockholders.

If applicable, the undersigned has the right to give voting instructions to Vanguard Fiduciary Trust Company, Trustee of the Alaska Air Group, Inc. Alaskasaver Plan, the Alaska Airlines, Inc. COPS, MRP and Dispatch 401(k) Plan, and the Horizon Air Industries, Inc. Savings Investment Plan, and/or Fidelity Management Trust Company, as Trustee of the Alaska Airlines, Inc. Pilots Investment and Savings Plan, at the Annual Meeting of Stockholders. This form, when properly executed, will be voted as directed. If voting instructions are not received by the proxy tabulator by 11:59 pm, Eastern Time on Thursday, May 12, 2011, your shares will not be voted by the Trustee.

The Annual Meeting of Stockholders is to be held at 2 p.m. on Tuesday, May 17, 2011, at the Museum of Flight, 9404 East Marginal Way South, Seattle, Washington, and at any adjournment or postponement thereof.

When this proxy is properly executed, the shares to which this proxy relates will be voted as directed. If no such directions are made, this proxy will be voted <u>FOR</u> the Election of Nominees for the Board of Directors, <u>FOR</u> the Board s Proposals 2, 3, and 6, <u>EVERY ONE YEAR</u> for Proposal 4, and <u>AGAINS</u>T Proposal 5.

Please mark, sign, date and return this proxy card promptly using the enclosed reply envelope or by voting over the internet or by telephone.

Continued and to be signed on reverse side.

C Non-Voting Items
Change of Address Please print new address below.

¢